ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Banerjee, Prasenjit; Iversen, Vegard; Mitra, Sandip; Nicolò, Antonio; Sen, Kunal

Working Paper Politicians and their promises in an uncertain world: Evidence from a lab-in-the-field experiment in India

WIDER Working Paper, No. 2019/60

Provided in Cooperation with: United Nations University (UNU), World Institute for Development Economics Research (WIDER)

Suggested Citation: Banerjee, Prasenjit; Iversen, Vegard; Mitra, Sandip; Nicolò, Antonio; Sen, Kunal (2019) : Politicians and their promises in an uncertain world: Evidence from a lab-in-the-field experiment in India, WIDER Working Paper, No. 2019/60, ISBN 978-92-9256-694-4, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2019/694-4

This Version is available at: https://hdl.handle.net/10419/211295

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WIDER Working Paper 2019/60

Politicians and their promises in an uncertain world

Evidence from a lab-in-the-field experiment in India

Prasenjit Banerjee,¹ Vegard Iversen,² Sandip Mitra,³ Antonio Nicolò,^{1,4} and Kunal Sen^{1,5}

August 2019

United Nations University World Institute for Development Economics Research

wider.unu.edu

Abstract: In emerging economies, pro-social policy outcomes may be prevented by bureaucratic inefficiency, capture by elected or non-elected office holders, or by other hurdles. For local citizens, uncertainty about the true cause of such failures often prevails. We study the pro-sociality of politicians' decision-making in a modified dictator game with real politician participants in rural India. In our game, a recipient citizen does not know whether dictator politician capture or bad luck is to blame when receiving zero. Using a 2×2 design, we investigate how the combination of two non-monetary instruments affect politician behaviour in this hard to govern environment. The first instrument, a (non-binding) promise, is a commitment device; the second introduces a minimal relational lever between the politician and the recipient. We find that politician-dictator giving becomes dramatically more pro-social, from zero to 50:50-giving, when these two instruments are combined. Our results provide new insights about the scope for norm-based, low-cost mechanisms to tackle governance-related asymmetric information challenges in developing country settings.

Key words: Asymmetric information, lab-in-the-field experiment, promise, politician, social preferences, India

JEL classification: H11, D64, O12

Figures and tables: at the end of the paper, all authors' work.

Copyright © UNU-WIDER 2019

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9256-694-4

https//doi.org/10.35188/UNU-WIDER/2019/694-4

Typescript prepared by Gary Smith.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland, Sweden, and the United Kingdom as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

¹ University of Manchester, Manchester, ² University of Greenwich, London; ³ Indian Statistical Institute, Kolkata; ⁴ University of Padova, Padova; ⁵ UNU-WIDER, Helsinki. Corresponding author: Prasenjit.banerjee@manchester.ac.uk.

This study has been prepared within the UNU-WIDER project Capable states-building the foundations for achieving SDGs

The United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

1 Introduction

Politicians are the same all over. They promise to build bridges even when there are no rivers. (Nikita Khrushchev)

While elected representatives fundamentally affect the welfare of citizens in a democracy, holding politicians to account is not always straightforward. In emerging economies, a pro-social policy may be held up by bureaucratic inefficiency, by the strategic conduct of elected or non-elected office holders, or by other institutional hurdles. In such settings, citizen monitoring of politician promises and behaviour (checks and balances) is made harder by the lack of information about what politicians and others are up to: even if a citizen knows what a representative has promised, it is difficult to ascertain whether a politician's actions or an institutional weakness is to blame if a pro-social policy fails. Uncertainty about the true cause—with a hard-tomitigate asymmetric information challenge at its core—will therefore often prevail.

The textbook solution to such a dilemma in a hard-to-govern environment is to introduce an incentive mechanism that seeks to align the interests of the politician and the electorate. In emerging economies, public purse constraints make such standard—and often costly—incentive schemes much less attractive. Viable alternatives should therefore be considered.

In this paper, we use a lab-in-the-field experiment in rural India to investigate the impacts of two norm-based (e.g. Krupka & Weber, 2013) and low-cost instruments on politicians' pro-social behaviour in the presence of asymmetric information that it is easy for politicians to strategically exploit. In line with the experimental literature, we think of the first instrument as a commitment device, a (non-binding) promise (Vanberg 2008); the second introduces a minimalist relationship between the politicians briefly greet their respective recipients before decision-making). Evidence from lab and field experiments shows how removing recipient anonymity affects generosity in dictator games (see Bohnet & Frey, 1999; Hoffman, McCabe, & Smith, 1996).

In our modified dictator game, we retain the hard-to-mitigate asymmetric information challenge as a core feature of our design by making it easy for a politician-dictator to blame a self-serving decision on factors outside their control (extending Andreoni & Bernheim, 2009). In the game, nature plays with a high probability (p = 0.8) and randomly assigns the endowment to the dictator or the recipient. A politician dictator plays with complementary probability (and knows, when making a decision, that this choice will be implemented): in contrast, a recipient who gets zero (or the full endowment), does not know whether the dictator or nature is responsible. In our 2×2 design, we keep p constant across treatments while varying our promise and minimalist relationship instruments to investigate their effect on politician response.

Our findings may be summarized as follows: (1) politicians keep their promises, and (2) the welfare effects for recipients are strongest when promise making and non-anonymity are combined. In our benchmark treatment a standard dictator game with an anonymous recipient—we observe high capture: politician-dictators allocate almost the entire endowment to themselves (average giving represents around 1% of the endowment, and 87% of politician-dictators give zero). When we remove anonymity and give the recipient a face, politicians become dramatically more generous: average giving rises to 33% of the endowment while zero-giving decreases to 27%. This minimalist relationship makes it costlier for politicians to behave greedily, in spite of the ease with which selfish behaviour can be concealed.

A non-binding promise significantly augments the pro-social decisionmaking of politician-dictators: in the anonymous setting, they promise a positive but small (i.e. 8% of the endowment on average) amount and keep their word. Crucially, politician behaviour changes dramatically from zerogiving in the anonymous, no-promise setting towards equal-sharing in the non-anonymous setting with promise (where politician-dictators greet their respective recipients and are forced to make a promise in front of them). Specifically, we observe that the proportion of equal sharing increases from 0% to 45%, while the frequency of zero-giving drops from 37% to 14%. Ninety per cent of politicians promise to distribute a positive amount and 81% keep their promise.

These findings are striking, first, because of the strong, pro-social politician response to simple experimental stimuli. This response occurs in an environment where it remains easy for politicians to blame selfish behaviour on factors outside their control. Given that laboratory experiments demonstrate that minor contextual features of a choice environment may lead to substantially different choices and outcomes (Krupka & Weber, 2013), our findings are consistent with a stable preference for complying with social norms. The findings are also notable because our politician participants are matched with ordinary citizen residents of villages at sufficient distance from their constituencies to ensure no contact in the past, and prior to the experiment, and no contact after the experiment is over.

Our research also improves the external validity of behavioural experiments (which often involve student participants) by examining the behaviour of hard-to-recruit real-world politicians. We recruit bottom-tier politicians (i.e. village council elected representatives) who have lower opportunity costs of time and serve relatively small number of constituents (approximately 3,100 persons, on average, per council, as reported by Anukriti and Chakravarty (2019)) for two Indian states—West Bengal and Uttar Pradesh. These politicians are elected through a conventional democratic process and are responsible for implementing a variety of government-funded development programmes and for decisions about investment in local infrastructure, such as sanitation, drinking water, and roads (Chattopadhyay & Duflo, 2004).

A simple model provides a theoretical grounding for our results. We assume that (some) dictators suffer a cost when breaking a promise and experience a moral reward when a promise is kept. All dictators enjoy a positive utility when they meet their recipients and give a generous amount (this can be loosely defined as social image), but differ in the extent that utility loss matters. Since such utility loss only arises in non-anonymous interactions, dictators offer zero in the anonymous game without a promise. When forced to make a promise in the anonymous setting, politician-dictators promise to distribute a small positive amount and keep this promise. The most interesting feature of the model arises for the non-anonymous interactions where image concerns enter the frame. With no promise, dictators with major image concerns distribute a fair amount, while dictators who are less concerned about their image can opt to distribute zero and hide their greediness behind nature. However, the prospect for hiding greediness behind nature is overridden when politicians are forced to make a promise. In this case, they will either be untruthful or disclose their greediness, by promising zero or a small amount. Dictators with a sufficiently high aversion to not keeping their word and a high concern for their image will therefore be forced to behave generously.

To provide an external validity check of our results, we test for social

norms of equal-sharing and/or promise-keeping among politicians, which our model assumes. Since survey data are vulnerable to social desirability bias concerns, we adapt Krupka and Weber's (2013) incentivized norm elicitation procedure using a simple coordination game. Politician respondents were asked to rank different allocations between the dictator and the recipient in a hypothetical dictator game scenario, and a combination of promise and give in a promise-game scenario (the dictator game with a prior non-binding promise) from 'very socially inappropriate' to 'very socially appropriate' on a four-point Likert scale. Respondents received monetary incentives to match the modal response of others in the same choice environment. The results are consistent with the findings of the experiment in that politicians believe (1) a distribution of between 40% and 50% of the endowment is socially acceptable, and (2) promise-breaking is socially unacceptable.

Our work provides a first attempt to investigate whether self-selected politicians have preferences for promise-keeping in a non-strategic environment (i.e. without electoral competition) and whether a minimal relational layer affects their behaviour. In our theory-grounded experimental design, a politician may break a promise and capture the entire endowment (a) because of factors genuinely outside the politician's control or (b) by exploiting uncertainty about the true state of the world. The novel feature of (b) is that the politician is well placed to blame (a) when a promise is not kept.

The combination of non-anonymity and a promise induce fair behaviour in politicians, even in the presence of private information about the true state of the world. These findings provide new insights to debates about the design of cost-effective mechanisms to prevent politician and other capture in environments where it is difficult to distinguish capture from other hurdles to pro-social outcomes: this is of special relevance in developing-country settings with decentralized systems of governance, where the use of incentive-based disciplining mechanisms is constrained by the higher opportunity costs of public funds.

The rest of the paper is organized as follows. In Section 2, we briefly discuss the background and related literature. Section 3 outlines the theoretical framework and spells out our main hypotheses. Section 4 presents the research design, including the game and experimental procedures. Section 5 presents the analysis and main findings from the experiment, Section 6 presents the design and main findings from the survey, and Section 7 concludes.

2 Related literature and motivation

Politicians are, perhaps, more familiar with promises than most others and the fulfilment of election promises features centrally both in the mandate theory of democracy and in the responsible party model (Besley & Coate 1997; Downs 1957; Osborne & Slivinski 1996). While stereotypes suggest that promise-breaking belongs to the fine art of political practice (e.g. ISSP Research Group 2008; Thomson 2011), it is also consistent with traditional choice theory,¹ which forms the backbone of the political economy literature (e.g. Besley & Coate 1997; Osborne & Slivinski, 1996). In contrast, recent experimental studies (using student subjects) suggest that people, in general, have social preferences for promise keeping (Ellingsen & Johannesson 2004; Vanberg 2008), while promise-breaking imposes an intrinsic psychological cost (Charness & Dufwenberg 2006; Charness & Gneezy 2008; Gneezy 2005; Hao & Houser 2010).² The two predominant views suggest that promises either induce emotional commitments to fulfil contractual obligations based on a norm of promise-keeping (Ellingsen & Johannesson 2004) or will be kept because of guilt from letting down the payoff expectations attributed to others (e.g. guilt aversion) (Charness & Dufwenberg 2006). Vanberg (2008) investigated whether promise-keeping is due to commitment preferences or guilt aversion, and found support for the former. Following Vanberg's (2008) argument, we test whether politicians have social preferences for keeping their promises and whether such commitment could help improve citizen welfare (i.e. higher giving by politician-dictators) in the presence of asymmetric information about the true state of the world.

Evidence from lab and field experiments document the importance of anonymity in explaining generosity in dictator games. For example, Hoff-

 $^{^{1}}$ In the absence of a mechanism or contract, an agent should always break a promise and extract rents from office if this is consistent with material self-interest (e.g. contract theory (e.g. Akerlof 1978), mechanism design theory (e.g. Holmstrom 1979)).

²A recent economics literature that draws on the social psychology theory of commitment suggests that non-binding promises and oath-taking significantly affect behaviour by increasing cooperation and coordination in social dilemmas and contributions to public goods (Carlsson et al. 2013; Jacquemet et al. 2013). Everyday instances of promisemaking include a truth-telling oath as part of court protocol and physicians being required to take the Hippocratic Oath before they start practising medicine. According to the social psychology of commitment, the oath or promise works as a 'preparatory act' (Burger 1999): compliance with an initial oath (or promise) requires changes in behaviour in subsequent decision-making situations.

man et al. (1996) find that reducing the dictator's anonymity results in more generous offers, and conjecture that a less anonymous experimental design evokes levels of strategic reciprocity common to everyday repeated social interactions. Bohnet and Frey (1999), however, argue that dictator generosity is driven not by reciprocity but by the ability to identify with the recipients, whether by knowing something about them or seeing their faces.³ Charness and Gneezy (2008) find similar evidence: recipients (located in a different city) identified by their family names receive significantly larger amounts. In the political economy literature, Barton, Castillo, and Petrie (2014) show that candidates' personal contact with voters in door-todoor canvassing is a more effective tool to persuade voters and increase the vote-share (see also Pons 2018). Also, according to the theories of spatial competition, direct information transmission of policy positions could alone influence voters' position (see Dewan & Shepsle 2011 for a review). Our paper explores the interaction between degrees of anonymity and a non-binding promise on politicians' pro-social behaviour in a controlled setting in rural India.

While the role of social preferences among politicians has been studied for several decades by political scientists and psychologists (see, e.g. Calvert 1985; Wittman 1983), the political economy literature has recently started exploring political competition with non-standard preferences of political actors and/or voters. For example, candidates can have heterogeneous motives, that is, along two dimensions: policy preferences and aversion to lying (Callander & Wilkie 2007), or may have a different 'character' (Kartik and McAfee 2007) or skills (Buisseret & Prato 2016). Alternatively, heterogeneity may stem from public spirit motives (altruism) or honesty (incorruptibility) (Bernheim & Kartik 2014). Moreover, candidates lacking such intrinsic motives can signal such unobservable characteristics strategically to voters to improve their reputation, which eventually helps them achieve ulterior selfinterested motives (Ariely et al. 2009; Béenabou & Tirole 2006; Callander 2008; Dana et al. 2007).

Substantial knowledge gaps remain about the respective importance of so-

³Extending this view, some recent literature has investigated the effect of social networks on giving in the dictator game. In particular, Leider et al. (2009) found that dictators give more to 'friends', i.e. recipients with social distance equal to 1, by combining network elicitation among college students followed by a controlled experiment. Goeree et al. (2010) concluded that dictators' giving is strongly dependent social distance: 'giving follows a simple inverse distance law'.

cial and moral motivations for political selection and politicians' behaviour since the extant literature is not well equipped to provide persuasive empirical support to efforts to tackle the disentangling of motivational factors. Put differently, it is hard to isolate one type of politician motivation from another, and observed politician behaviour is unlikely to accurately guide such research efforts. While empirical research is well positioned to successfully document politician competence (e.g. their education or legislative efforts) (e.g. see Dal Bó et al. 2017; Ferraz & Finan 2011), the empirical study of politician motivation involves tougher identification challenges.

In response, and aided by incentive-compatible mechanisms, economists step back to the lab to disentangle motivations within controlled settings. In lab experiments with student subjects, Corazzini et al. (2014) allowed candidates to make promises to voters about how they will divide the pie that the election winner receives between themselves and the voters, and found that campaign promises may not be purely cheap talk (also see Corazzini et al. 2007; Geng et al. 2011).

For the present inquiry, standard lab experiments with student participants are of limited value since selection into politics and academic studies are incomparable. Some recent studies therefore conduct controlled experiments with real politician participants. Using experiments and regression discontinuity with elected and non-elected real politicians in Zambia, Enemark et (2016) investigate whether reciprocity, clientelism politics, or corrupt al. exchange are inherent characteristics of self-selected politicians, or acquired while in office. They find a moral-hazard rather than adverse-selection explanation, as holding office increases reciprocal behaviour. Kosfeld and Rustagi (2015) explore the existence and impact of social preferences among selfselected leaders on common-pool resource management by combining results from a controlled experiment with data on leader performance. In their social dilemma experiment in which leaders could punish group members for anti-social behaviour, altruistic leaders (who care about equity and efficiency in the controlled setting) promote better common forest management outcomes (also see Jack & Recalde 2015). No existing study explores politicians' pro-social decision-making in the type of hard-to-govern environment that we attempt to mimic here.

Our work also contributes to the growing literature on social preferences and self-selection into jobs. There is ample recent evidence in the empirical and experimental literature that individuals self-select into occupations and organizations that better match their needs and aspirations (e.g. Banuri & Keefer 2016; Besley & Ghatak 2005; Delfgaauw & Dur 2007; Hanna & Wang 2017). Politics is a highly unusual occupation and those who enter it may do so not only because they expect to extract a rent from their time in office, but also because they are motivated by a sense of public duty, altruism, or by status and prestige. Politicians' motivations vary across societies, cultures, times, and places, and change with institutional settings (Beniers & Dur 2007; Besley 2005; Braendle 2016; Fedele & Naticchioni 2016; Gavoille & Verschelde 2017).

3 A theoretical framework

We analyze four different versions of a simple dictator game in which a dictator (D) and a recipient (R) split a prize normalized to have a unit value. Let $x \in [0, 1]$ denote the transfer R receives; D consumes 1 - x. The timing of the game is as follows. Nature plays first: with probability $1 - \pi$ the prize is randomly assigned either to R or to D, and with probability $\pi > 0$, Ddecides how to split the prize and the game ends. The probability $\pi \in (0, 1)$ is common knowledge, but R cannot observe whether nature intervened. The variation among games consists in whether D makes a non-binding promise to R before nature plays or not, and whether the interaction among D and R is anonymous or they meet each other before playing the game. We assume the the dictators are heterogeneous in their motivations, and hence such variations in the games may affect dictators' behaviour. A dictator i's utility function can be represented as

$$U_i(x,p) = 1 - x + \mathcal{I}^p \phi_i(p,x) + \mathcal{I}^a \tau_i(\widehat{x}^R(x|p), \overline{x}_i, \pi).$$
(1)

The first component is the material utility in consuming 1-x; the second component, the function $\phi_i(p, x)$, reflects both the costs and benefits of, respectively, breaking and fulfilling a promise, and is multiplied by an indicator function $\mathcal{I}^p \in \{0, 1\}$ that takes the value of 1 if the dictator makes a (nonbinding) promise before splitting the prize, and 0 otherwise. The third component, the function $\tau_i(\hat{x}^R(x|p), \overline{x}_i, \pi)$, reflects image-concern; where $\hat{x}^R(x|p)$ denotes R's beliefs about the amount offered by D when he receives x: a dictator i cares about her image, judged by the extent to which the amount offered, according to R's beliefs, departs from a reference point \overline{x}_i . In the case that the recipient receives either the full prize or zero, R forms beliefs about the amount that the dictator has offered, which depends on π (the probability according to which nature intervenes), and the promise made by the dictator. Concerns for image only arise in a non-anonymous interaction between a dictator and a recipient, and therefore also this component is multiplied by an indicator function $\mathcal{I}^a \in \{0, 1\}$ that takes value 0 if the interaction is anonymous and 1 otherwise.

We make a series of restrictive assumptions that allow us to identify a simple but meaningful setup. Dictators are of three types: (1) *h*-type dictators, who feel a utility loss if receivers believe that they have not distributed the prize fairly and suffer a utility loss from breaking a promise (and a utility gain in fulfilling it); (2) *ls*-type dictators, who feel a utility loss if receivers believe that they are greedy, but their reference split is lower, equal to x - t with t > 0,⁴ and they also are 'sincere', that is they suffer a utility loss from breaking a promise; and (3) *ld*-type dictators, who also have a reference split equal to x - t, but they are 'deceitful', in the sense that they do not suffer any utility loss from breaking a promise. Formally, a *h*-type dictator has ex-post utility

$$U_{i}^{h}(x,p) = 1 - x + \mathcal{I}^{p} \alpha \sqrt{p} - \mathcal{I}^{a} L \max\left\{ \left(\frac{1}{2} - \hat{x}^{R}(x|p)\right), 0 \right\}; \qquad (2)$$
$$p \le x 1 - x - \mathcal{I}^{a} L \max\left\{ \left(\frac{1}{2} - \hat{x}^{R}(x|p)\right), 0 \right\} - C; p > x$$

with L > 1 and C > 1. The parameter $\alpha \in (0, 1]$ captures the intrinsic benefit of a sincere dictator for fulfilling a promise which is increasing and concave in the amount promised. Note that a sincere dictator's cost and benefit of fulfilling or breaking a promise depend on his own intrinsic preferences and not on R's opinion about D's behaviour and therefore they are relevant also in anonymous games.

An *ls*-type dictator has an ex-post utility equal to

⁴Here we assume that the 'reference' split for *h*-type dictators is equal to $\frac{1}{2}$, but more in general we could have assumed that is equal to $\frac{1}{2} - t^h$ with $t^h \ge 0$ and $t^h < t^l$. Results would change accordingly.

$$U_i^{ls}(x,p) = 1 - x + \mathcal{I}^p \alpha \sqrt{p} - \mathcal{I}^a L \max\left\{ \left(\frac{1}{2} - t - \hat{x}^R(x|p)\right), 0 \right\}; \quad (3)$$
$$p \le x - \mathcal{I}^a L \max\left\{ \left(\frac{1}{2} - t - \hat{x}^R(x|p)\right), 0 \right\} - C; p > x$$

where the parameter t determines a reference split, departing from which an *ls*-type dictator suffers a constant marginal loss of utility equal to L > 1if he is considered a greedy individual. We assume that $\frac{1}{2} > t > 0$. Finally, an *ld*-type dictator has an ex-post utility equal to

$$U_i^{ld}(x,p) = 1 - x - \mathcal{I}^a L \max\left\{\left(\frac{1}{2} - t - \hat{x}^R(x|p)\right), 0\right\}$$
(4)

Notice that a *h*-type dictator suffers a utility loss when *R* believes he has not split the prize equally. The other types of dictators suffer a loss when *R* believes they have distributed less than $\frac{1}{2} - t$.

We aim to predict how much dictators distribute in each of the following games:

- 1. an NPA (no promise, anonymous) game in which the interaction is anonymous and D cannot make any promise $(\mathcal{I}^p = \mathcal{I}^a = 0);$
- 2. a PA (promise, anonymous) game in which the interaction is anonymous and D has to make a promise at the beginning of the game, before knowing whether nature intervenes ($\mathcal{I}^p = 1$; $\mathcal{I}^a = 0$);
- 3. an NPNA (no promise, non-anonymous) game in which the interaction is non-anonymous and D cannot make any promise $(\mathcal{I}^p = 0; \mathcal{I}^a = 1);$
- 4. a PNA (promise, non-anomymous) game in which the interaction is non-anonymous and D has to make a promise at the beginning of the game, before knowing whether nature intervenes ($\mathcal{I}^a = \mathcal{I}^p = 1$).

In PA and PNA games, dictators play sequentially, making a promise before nature intervenes and then splitting the prize if they have to make the decision. For this reason, we use the subgame perfect Nash equilibrium concept and we refer to it simply as 'an equilibrium' of a game. Recipients are not players, but since their beliefs affect dictators' payoff, they are relevant in determining dictators' equilibrium strategies. We impose standard requirements on recipients' beliefs: if D distributes $x \in (0, 1)$, R's beliefs about how much D has distributed are trivially equal to x. When observing x = 0, R's beliefs should be consistent, that is, they are computed according to equilibrium strategies and Bayes' rule whenever is possible. As usual, arbitrary beliefs out-of-equilibrium path may induce a multiplicity of equilibria. For this reason, we impose a very mild behavioural assumption, that is beliefs satisfy a weak monotonic requirement: if $\hat{x}^R(x|p) = x$ then for all p' > p, $\hat{x}^R(x|p') \ge x$ and $\hat{x}^R(x|p > \frac{1}{2}) = \hat{x}^R(x|p = \frac{1}{2})$. It is important to point out that we do not impose that a larger promise strictly increases R's beliefs, but that a larger promise cannot worsen R's beliefs. We first analyse anonymous games (games 1 and 2 above), in which $\mathcal{I}^a = 0$ and therefore dictators' image concerns do not play any role. In PA and NPA games dictators do not make a promise before splitting the prize, and therefore we simplify the notation of R's beliefs and we write $\hat{x}^R(x)$, where x is the amount distributed.

The following proposition illustrates dictators' equilibrium strategies in these games.

Proposition 1 Consider an NPA game. In equilibrium every dictator distributes an amount equal to zero. Consider a PA game. In equilibrium sincere dictators fulfill their promise, and promise an amount equal to $\frac{1}{4}\alpha^2$; deceitful dictators distribute zero.

Proof: see the Appendix.

In an NPA game, dictators behave selfishly: when $\mathcal{I}^p = \mathcal{I}^a = 0$, dictators maximize their material utility. In PA games, sincere dictators are motivated by the intrinsic moral benefit of fulfilling a promise. Since deceitful dictators do not suffer any cost from breaking a promise, they do not get any benefits in fulfilling it—they distribute zero. Any promise by a deceitful dictator can be part of an equilibrium.

We focus now on non-anonymous games in which dictators care about R's beliefs about the amount they offer. We first analyze the NPNA game in which dictators do not make any promise. We refer to l-type dictators when referring to ls-type and ld-type because their behaviours coincide in games without promises.

Proposition 2 Consider an NPNA game. In equilibrium, h-type dictators distribute a positive amount equal to $\frac{1}{2}$; l-type dictators distribute zero if

1.
$$t \geq \frac{1}{2}\left(1 - \frac{\gamma(1-\pi)\frac{1}{2}}{(1-\gamma)(\pi+\frac{1}{2}(1-\pi))+\gamma(1-\pi)\frac{1}{2}}\right)$$
, or
2. $t < \frac{1}{2}\left(1 - \frac{\gamma(1-\pi)\frac{1}{2}}{(1-\gamma)(\pi+\frac{1}{2}(1-\pi))+\gamma(1-\pi)\frac{1}{2}}\right)$ and $L \leq \frac{\frac{1}{2}-t}{\frac{1}{2}-t-\hat{x}_{0}}$;
l-type dictators play a mixed strategy offering zero with some positive probability and $\frac{1}{2} - t$ with complementary probability.

The intuition of this result is the following. If a dictator offers a positive amount, she offers her reference split: $\frac{1}{2}$ for a *h*-type and $\frac{1}{2} - t$ for an *l*-type. If all dictators offer zero then $\hat{x}^R(0) = 0$, but this cannot be an equilibrium because, in this case, a *h*-type *D* would profitably deviate offering $\frac{1}{2}$. It follows that in every equilibrium *h*-type dictators offer $\frac{1}{2}$. Consider a strategy profile such that *h*-type dictators offer $\frac{1}{2}$ and *l*-type dictators offer zero. If $\hat{x}^R(0) \geq \frac{1}{2} - t$, because there are many *h*-type dictators and nature plays with sufficiently high probability, then this strategy profile is an equilibrium. Offering zero is clearly a best response for *l*-type dictators because they do not suffer any utility loss due to a bad image. If $\hat{x}^R(0) < \frac{1}{2} - t$, offering zero is a best response only if the utility loss due to a bad image is not too large, that is, only if $1 - L(\frac{1}{2} - t - \hat{x}^R(x)) \geq \frac{1}{2} + t$. If the previous condition does not hold, then in equilibrium *l*-type dictators play a mixed strategy that makes them indifferent between offering $\frac{1}{2} - t$ and offering zero.

Consider finally a PNA game. Now the behaviour of *ls*-type and *ld*-type dictators may differ because the former suffer a utility loss if they break a promise.

Proposition 3 Consider a PNA game. In equilibrium, h-type dictators promise and distribute a positive amount equal to $\frac{1}{2}$; ls-type dictators promise and distribute a positive amount equal to $\frac{1}{2} - t$; ld-type dictators promise $\frac{1}{2}$, and offer zero if either $\widehat{x}^R(0|p = \frac{1}{2}) \geq \frac{1}{2} - t$ or $1 - L(\frac{1}{2} - t - \widehat{x}^R(0|p = \frac{1}{2}) \geq \frac{1}{2} - t$, otherwise they offer $\frac{1}{2} - t$.

Proof: see the Appendix.

When dictators have to make a promise before splitting, ls-type dictators cannot hide behind nature. If they promise a positive amount, then they are forced to fulfill it. If they promise zero, they suffer a utility loss due to a very bad image. Since h-type dictators promise what they offer, that is $\frac{1}{2}$, ls-type dictators are forced to promise and distribute $\frac{1}{2} - t$; ld-type dictators mimic the promise of h-type dictators and offer zero if recipients' beliefs are large enough, that is, if the number of h-type dictators and the probability that nature moves are sufficiently large, otherwise they offer their reference split. The previous analysis can be summarized in the following simple predictions on dictators' behaviour in the four games.

Prediction 1: In an NPA game dictators distribute zero.

Prediction 2: In a PA game, sincere dictators distribute a (small) amount equal to $\frac{1}{4}\alpha^2$.

Prediction 3: In an NPNA game dictators with high concern for their image offer $\frac{1}{2}$. Dictators with lower concern offer zero if recipients who get zero believe with sufficiently large probability that nature has intervened.

Prediction 4: In a PNA game dictators with high concern for their image promise and offer $\frac{1}{2}$. Dictators with lower concern who suffers a disutility in breaking their promise will promise and offer $\frac{1}{2} - t$; deceitful dictators promise a positive amount but offer zero if recipients who get zero believe with sufficiently large probability that nature has intervened.

4 Recruitment and experimental design

We envisaged two main organizational challenges in recruitment: (1) recruiting real politicians as subjects; and (2) creating a neutral field-lab environment.

For recruitment, we take advantage of India's decentralized and democratic local governance structure, the Panchayat system. This system has three tiers: Gram Panchayat (village-level councils), Panchayat Samiti (blocklevel councils), and Zila Parishad (district-level councils). A Gram Panchayat is divided into Samsads (wards). Citizens elect representatives for each tier, starting from Samsads, and elections are held at regular, five-year intervals.⁵ Village-level elected representatives generally do not have a role in the higher tiers (e.g. block or district level) unless they are the village council head.

Through the 73rd Constitutional Amendment (1993), village councils were given responsibility for implementation of a variety of governmentfunded development programmes and decisions about investments in local

⁵The politicians at the bottom tier of this system (Samsad or ward leader) represent around 500–800 voters (around 200–300 households) and are members of a village council or GP. GPs usually serve around 3,000-5,000 voters, although size varies widely. The second tier (i.e. block level) consists of 10–12 GPs and the final tier is the district council (i.e. Zila Parishad), which consists of 15–20 (on average) blocks.

infrastructure, such as sanitation, drinking water, and roads (Chattopadhyay & Duflo 2004). The elected representatives of interest here can thus exercise considerable power in their constituencies.

Our definition of a politician is a person who has either recently fought or recently won (in the last 10 years) an election for a village council (Gram Panchayat or GP) seat as a ward member.⁶ These self-selected politicians' preferences—whether selfish or social—have not been studied in depth. Monetary incentives for holding office are limited (the official salary of the village head is about USD 50/month; ward leaders are paid even less), but there are potential private returns from political rents and corrupt practices.⁷

Hooghly district in West Bengal and Varanasi district in Uttar Pradesh were selected because of prior experience working there. From among the administrative blocks in each district, we randomly selected two blocks following a stratified random sampling based on geographical location. For example, from among Hooghly's 18 administrative blocks, we randomly selected Singur and Dhaniakhali. In Uttar Pradesh, Badagaon and Sevapuri blocks were selected using a similar procedure. GPs were then randomly selected from each block. For each GP, we prepared a list of individuals who had contested or been elected during the two most recent elections and invited politicians to participate with an invitation letter prepared by the research team following a blinded, random protocol. The letter neutrally framed the purpose of the study (e.g. we want to study the challenges of rural development) and explained the random selection of the village/GP and participants (see letter text in the supplementary Appendix). Participants knew that they could change their decision to participate any time without giving any explanation. We made sure that politicians and other participants from one GP should not have any prior knowledge about their matched counterparts

⁶We purposely avoided recruiting village council heads (pradhans) because of their typically greater and more visible role in their party's political machinery, and their higher likelihood of being known to more villagers within a district, including among those from distant locations. The opportunity cost of time for village council heads would also be higher than for ward members.

⁷Some evidence suggests that an average candidate spends USD 400–800 during a village council election (see: www.ndtv.com/india-news/the-rs-81-500-crore-lie-565175). The average declared wealth of re-contesting candidates to Parliament and state legislative assemblies in 2004 was 134% higher than during the first election (Sastry 2014), suggesting high rents. Fisman et al. (2014) also show that the annual asset growth of winners in state elections is 3–5 percentage points higher than for runners up. Although similar statistics are not available for village council candidates, the returns are likely to be non-trivial.

from another GP. We also chose the timing of the experiment carefully to avoid any overlap with election-related or other political campaigning (we conducted our experiments from December 2016 to April 2017).

From each village and based on the household census, we also invited randomly selected ordinary citizens (non-politicians) to participate in the experiment. The presence of non-politicians aimed to reduce experimental demand effects, since a sample comprising only of politicians could intensify the feeling of being under experimental scrutiny.

Our research assistants recruited local enumerators to collect participant information. They prepared a list (census) of households, which was always kept with them only, containing basic demographic information (name of household head, sex, education, occupation). Following a blinded, random protocol, the enumerators selected potential participants and invited them, following the same procedure described above (i.e. there were no differences in the recruitment/invitation procedure (e.g. same draft letter used in both cases) used for politicians and non-politicians). Participants knew they had to travel 25 km on average to participate in the study and play a game, but they had no prior information about where they were going and with whom they would play the game. As participants had to travel long distances to participate and there was no well-connected public transport available, we arranged free transport for them, as well as refreshments.

4.1 Design

We use a 2×2 design described in the table below.

	Anonymous	Non-anonymous
No promise	T1 (NPA)	T2 (NPNA)
Promise	T3 (PA)	T4 (PNA)

We carefully explain the procedure followed in each treatment in the Appendix. Here, we provide a summary of the most relevant features of the design.

Forty subjects from two different villages participated in each session, one in the home-village (where the venue was located) and one in the visitorvillage (a distant location): 10 politicians and 10 non-politicians participated from each village. Participants were randomly assigned to the role of dictator (D) or receiver (R) and kept their role during the entire session. Each subject was randomly matched with a subject from a different village. As noted, the distance between the home-village and the visitor-village is sufficient to ensure that subjects from home- and visitor-villages are likely to have never met before the experiment and will not meet again after.

In the 'anonymous' treatments T1 and T3, there were 20 participants: 10 politicians and 10 villagers in each room, all from the same village. They were matched with participants from a distant village sitting in a different room. In the 'non-anonymous' treatments (T2 and T4), 10 participants from the home-village and 10 participants from the visitor-village (randomly chosen) were seated in the same room. Each pair D and R (from different villages) were asked to stand up and greet each other. In anonymous treatments matched group members do not meet each other. In the non-anonymous treatments, each group member stands up and greets the other before the game starts.

Each pair received a fixed and known endowment—1000 INR (approx. USD15.50)—for each round and D had to decide how to allocate the endowment between him/herself and his/her partner (i.e. R).

Each D received a random (and confidential) private number between 1 and 10—no other person in the room, not even the experimenter, would know this number. At the start of each round, the experimenter announced two randomly chosen numbers between 1 and 10. Each D received a decision sheet. They filled in their decision sheets in an enclosed area one by one. Only Ds whose private numbers were announced could choose and record a distribution on the decision sheet in private, others would just tick a box which stated that nature would give zero to either D or R (see the decision sheet example in the Appendix). All Ds who made a decision or who ticked a box folded the decision sheets and put them in an envelope, named, for example, Round 1-Decisions, themselves. The Rs, other Ds, and the experimenter knew the probability (i.e. 0.8), but did not know whether nature or D made the decision when the outcome was either zero or 1000 INR (this can only be true if D chooses the same division as nature).

In the promise treatments (i.e. treatment T3 and T4), all Ds wrote how they would allocate INR 1000 between him/herself and R on a 'promise slip', before the two numbers were announced, and therefore before knowing whether D or nature would decide how to divide the endowment. Each D went to an enclosed area and wrote this in private; then they put the folded promise slip into an envelope and returned it to the experimenter. The experimenter passed it (without looking at it) onto the respective R (sitting in the same room in T4 or sitting in another room in T3, in which the experimenter carried the promise slip to the corresponding Rs). Each R observed what his/her partner promised to give him/her in private and then put them in an envelope named, for example, Round 1-Promise. Then we followed the dictator game in the anonymous and non-anonymous settings described above.

We repeated each treatment five times. At the end, one of the five rounds was randomly selected to determine the payments. The envelope of decision sheets for the selected round was handed to a person outside the venue (external person) who had no information about the game or the subjects. He checked the decision sheets in a separate room and put the payment in a separate envelope for each dictator. He also decided whether D or R got INR 1000 when nature intervened by flipping a coin. The external person gave a result sheet to the experimenter, who then published the results. Each participant left the room one by one and received their envelopes with their payments (their earnings from the game plus a fixed participation fee of INR 300) from the external person, based on the decisions they or their partners or nature made. They left the venue one by one. Participants from the visitor-village left the venue first.

5 Analysis

Our lab-in-the-field experiment sample contains 175 politicians and 133 nonpoliticians. Table A1 presents the summary statistics of the observable characteristics of politicians, by gender, educational level, age, caste, and occupation. We note the following: (1) 34% of politicians are female; (2) politicians have 9.6 years of education (on average); (3) their average age is 40.5 years; (4) 44% of politicians are from a forward caste background; and (5) about 50% of politicians are farmers.

We find that: (1) politicians keep their promises, and (2) the welfare effects are strongest when promise-making and non-anonymity are combined. Our politicians behave selfishly (i.e. mostly give nothing) in the anonymous dictator game but behave pro-socially (i.e. giving 33% of the endowment on average) in the non-anonymous treatment in which they meet and greet their respective recipients before making a decision. In the anonymous setting, politicians promise a positive, but small (i.e. 8% of the endowment on average), amount and keep their promises. Most importantly, politicians' behaviour changes drastically from almost zero-giving in the anonymous nopromise setting to equal-sharing (and hence citizens' welfare is maximized) in the non-anonymous with promise setting (when they meet and are forced to make a promise in front of the recipients). Among our politicians, there is a social norm of (1) equal-sharing; and (2) promise-keeping.

We report the detailed results below.

Result 1: Politicians give significantly more in the non-anonymous dictator game (T2) compared to the anonymous dictator game (T1). There is a significant decrease in the proportion of zerogiving and a significant increase in the proportion of 50:50 giving in T2 compared to T1.

Following the first prediction of the model, in the first treatment (T1), politicians mostly give nothing: 87% of them give zero and average giving is Rs.13.33, or about 1% of the endowment. This suggests that our politicians are guided by self-interested preferences. When a minimalist relationship is introduced and politicians meet and greet their respective recipients before making a choice, we observe significant changes in politicians' behaviour they become more pro-social, as shown in Table 1. Compared to T1, in T2, average giving increases (from 1% to 35% of the endowment (i.e. from Rs.13 to Rs.356)), zero-giving falls (from 87% to 26.4% of cases), and equalsharing increases (from 0% to 32% of cases): each one of these changes is statistically significant (see Tables 1 and 2). This pro-social behaviour is also evident from Figure 1(a), which shows a kernel density of the amount given by politician-dictators in T1 and T2 (also see Figure 2, which displays the frequency of amount given across different treatments). As our theoretical framework predicts (Proposition 2 and Prediction 3), there are h-type, who give 50:50, and l-type, who give either zero or a positive amount less than 50:50, dictators in our sample. This suggests that adding a face acts as a catalyst to promote pro-social behaviour (as argued by Bohnet and Frey 1999) among politicians even when (1) uncertainty of the true state of nature prevails; and (2) politicians will not meet the 'unknown' recipients in the future.

Result 2: Politicians keep their promise.

Politicians are highly likely to keep their promise, both in the anonymous (T3) and non-anonymous (T4) setting. In T3, they keep their promises in

96% of cases. On average they promise Rs.87.33 and give Rs.80 in T3. In T4, they keep their promises in 81% of cases and, on average, they promise Rs.519 and give Rs.460. Our politicians are mostly sincere. The amounts promised and given in T4, however, are significantly larger than those in T3. Table 3 presents the amount promised and the difference between amounts given and promised, by the level of the amount promised, in T3 and T4. In T3, only 1 out of 30 politicians breaks her promise—she promises to give Rs.300 but actually gives nothing. In T4: (1) 53.2% of politicians promise exactly half of the endowment; (2) they keep their promise with the mean difference between amount given and amount promised being Rs.-20.9 for the whole T4 sample and Rs.6.1 for those who promised 50% of the endowment. Finally, 81% of politicians give the amount promised and 8% actually give more than they promise (Table 4). This promise-keeping behaviour is also shown in Figures 3 and 4. Figure 3 shows kernel densities of promise and amount given by politicians in T3 and T4. In Figure 4, we plot the amount promised against the amount given in T3 and T4. If dictators keep their promises, observations would be along the 45-degree line. For politicians it is evident that the line of best fit is very close to the 45-degree line (exact 45-degree line with one outlier in T3).⁸

A non-binding promise significantly affects politicians' behaviour: they become more pro-social and give more to recipients. However, recipient welfare does not improve as much as for the first mechanism (i.e. T2). In T3, politicians start promising non-zero amounts and keep their promise, but they give small amounts. This supports the Prediction 2 in our theoretical framework. The proportion of zero-giving drops significantly from 87% to 37% (see Tables 1, 2, and 3, and Figure 2). However, average giving is small and increases from 1% (in T1) to 8% (in T3) of the endowment (i.e. Rs.13 in T1 to Rs.80 in T3): recipient welfare does therefore not improve as much as for the first mechanism (Rs.356 in T2 vs Rs.80 in T3) (see Figures 1(d) and 3(a)). Figure 3(b), which is a kernel density plot of giving in T1 and T3, also supports this.

Result 3: Introducing a non-binding promise in the non-anonymous

⁸A regression of give on promise, controlling for other characteristics, shows a coefficient estimate being 0.82 (significant) and 0.89 (significant) and intercept being 66.19 (insignificant) and 20.72 (insignificant) in T3 and T4 respectively. A joint hypothesis that the slope = 1 and the intercept = 0 cannot be rejected in both the cases. This suggests the fitted lines between promise and give in T3 and T4 are not significantly different from a 45-degree line.

setting (T_4) makes politicians promise and give a fair allocation, 50% of the endowment.

The most striking results are observed in T4, when our two instruments, the minimalist relationship and promise, interact—politicians move from zero-giving to 50:50-giving. Promise has greater impact on politicians' prosocial behaviour when they are forced to make a promise in front of their respective recipient (they could still hide their actions by exploiting private information about the true state of nature). Results support Prediction 4 that compared to T3, we observe significant changes in behaviour in T4 average giving increases (from 8% (Rs.80) to 46% (Rs.460) of the endowment), the proportion of equal sharing increases (from 0% to 45% of cases), and the frequency of zero-giving drops (from 37% to 14% of cases) (see Tables 1 and 2, and Figure 2). Figure 1(f) shows the kernel density of giving in T3 and T4, which supports this observation. Following Prediction 4, our results suggest that most of the politicians in our sample are *h*-type (at least 45%), although we have *ls*-type and *ld*-type (around 15% each) politicians too.

Comparing results in T2 and T4, politicians are significantly more generous in T4 than in T2, with a shift to a 50:50 split. Results show a 31% increase in the mean amount given—with mean giving increasing from Rs.356.60 to Rs.459.67: this change is statistically significant (see Figure 2 and Tables 3) and 4). In T4, the proportion of politicians giving zero also drops notably from 26.4% to 14.5%. Ninety per cent of politicians promise to distribute a positive amount and 81% keep their promise (see Figures 3(b), and 4(b)). The kernel density plot for the amount given in T4 as compared to T2 shows a spike at the 50:50 split, suggesting more politicians move to a 50:50 sharing norm in the promise treatment with non-anonymity (Figure 1(e))—32% of politicians give 50:50 in T2 as compared to 45% in T4. Even if politicians could take the entire endowment without ruining their social image (as they can still hide their actions), they choose to be fair. These effects are striking, since we have taken care to ensure that local politicians are matched with recipients from distant villages that they are highly unlikely to have had any past or will have any future interaction with once the experiment is over.

We next investigate whether this difference in behaviour across treatments remains evident after controlling for observable politician characteristics. In separate regressions, reported in columns (1)-(3) of Table 5, we regress the amount given, and a dummy for zero-giving on the treatment dummy (equals 1 if promise treatment, 0 if no-promise treatment; similarly other treatment dummies) and politician characteristics (gender, age, and educational level along with dummies for occupation, caste, and the states where politicians were recruited from).

We find that the coefficients on the non-anonymous treatment (T2) and promise-non-anonymous treatment (T4) are significant at the 5% level when the dependent variable is the amount given, and with the right sign (controlling for their observable characteristics) politicians give significantly more in T2 and T4. However, while the sign of the coefficient for the promiseanonymous treatment (i.e. T3) is positive, as hypothesized, it is not statistically significant. We also find that zero-giving drops significantly in all treatments (column (2)). We also run separate regressions (ordinary least square or OLS) comparing two treatments at a time (see Table 6), with the results supporting the observation that the amount given is significantly higher across the treatments and zero-giving decreases significantly across the treatments, except when we compare T2 with T4.

Comparing the non-anonymous treatments without and with promise (i.e. comparing T2 and T4), the number of politicians who distribute zero decreases, but this number does not turn out to be significant. Our experimental results suggest that the fraction of politicians who respond to non-anonymity is high (and the fraction of intrinsically inequity-averse politicians is low) and that the welfare effect from removing anonymity is significant. The non-binding promise significantly improves citizens' welfare when implemented in the non-anonymous setting: politicians start to promise a fair distribution since they cannot hide and are forced to make a promise in front of the respective recipient. They keep this promise because of their preference for keeping their word.

Using the data on non-politician-dictators we can also make preliminary inferences about the behaviour of non-politicians.

Result 4a: Removing anonymity has an impact on non-politicians' behaviour: non-politicians are significantly more generous in T2 than T1.

Result 4b: Compared to politicians, non-politicians are less likely to keep their promise.

Result 4c: Compared to politicians, a non-binding promise is less likely to affect non-politicians' behaviour.

Non-politicians respond to non-anonymity and behave significantly more pro-socially in T2 than T1: (1) average giving increases from Rs.5 to Rs.335; (2) the proportion of zero-giving drops from 95% to 19.35% of cases; (3) the proportion of equal-sharing increases from 0% to 35.48% (these changes are all statistically significant, see Tables 9, A2, and A3). We also find that non-politicians are less likely to promise a positive amount and to keep their promises, compared to politicians, in the anonymous dictator game with promise (i.e. T3)—see Figure A2 and Tables 9, A4, and A5. In T3, they keep their promises, but the average amount given is almost zero (Rs.41). When they are forced to make promises in front of their respective recipient (i.e. T4), they start promising and giving significantly more. While 54.8% of nonpoliticians in T4 keep their promises, this is a significantly lower proportion of promise-keeping than among politicians, which is reflected in a line of best fit significantly different from the 45-degree line (Figure A2). We also find that there is very little difference in the allocation decisions of non-politicians in T2 and T4 (when promise is introduced in the non-anonymous setting) (Tables 9, A3, and A2)-35.48% of non-politicians gave Rs.500 in the nopromise treatment, which actually increased slightly to 35.7% in the promise treatment (with the mean amount given in the no-promise (T2) and promise treatments (T4) being Rs.335 and Rs.402, respectively). Further, the test statistics on differences in amount given, zero-giving, and 50:50-giving across T2 and T4 lack statistical significance (Table A2).⁹

Following the similar procedure used for the politician sample, we investigate whether this difference in behaviour across treatments remains evident after controlling for observable non-politician characteristics. We find that the coefficients on T2 and T4 are significant at the 5% level when the dependent variable is the amount given, and with the right sign—controlling for their observable characteristics—non-politicians give significantly more in T2 and T4 (see Table A6). We also find that zero-giving drops significantly in all treatments (column (2)).

We also test whether politicians and non-politicians differ in their promisekeeping, controlling for observable characteristics of politicians and nonpoliticians. We confine the sample only to the non-anonymous-promise (i.e. T4), and construct a dummy for promise-keeping, where the dummy is 1 when the amount given equals the amount promised, and 0 otherwise. We

⁹This is not a robust finding but simply a reflection of a small sample size and lack of statistical power.

estimate a probit regression where the explanatory variable of interest is a dummy for politicians, after pooling the politician and non-politician samples while controlling for the dictator's caste, gender, age, education, income, and location. We present the results in column (3) of Table 5. We find the coefficient on the politician dummy to be positive and statistically significant at the 1% level. This supports the idea that politicians are more likely to keep their promise than are non-politicians.

Thus, the patterns in the data suggest that there is no evidence of promises playing a similar commitment role for non-politicians. This is consistent with our theoretical predictions: ordinary citizens in our sample are either selfish or inequity-averse individuals, and only among politicians is there a non-negligible number of reputation-concerned individuals.¹⁰

6 Survey

We assume in our model and in the analysis that promise-keeping preferences are intrinsic (in other words, promise-breaking imposes a pure moral/ psychological fixed cost) and not driven by social-image concerns (i.e. shame or reputational costs from promise-breaking; see Cohen et al. (2011, 2012) for some psychological studies on shame and guilt).¹¹ This argument is consistent with the existing literature suggesting that people have social preferences to fulfil contractual obligations based on the norm of promisekeeping (Ellingsen & Johannesson 2004; Vanberg 2008). To test whether self-selected politicians obey norms of equal-sharing and promise-keeping, we adapt Krupka and Weber's incentivized survey using simple coordination games to elicit social norms. In the survey, politician respondents were asked to rank different allocations between the dictator and the recipient in the dictator game scenario and a combination of promise-give in the promise game scenario from 'very socially inappropriate' to 'very socially appropriate' on a quantified scale of one to four. Respondents received monetary incentives to match the modal response provided by others in the same choice environment

¹⁰The reason why only among politicians there is a sizeable proportion of reputationconcerned individuals could be either due to self-selection into politics by those types or due to an attitude induced by their status: the answer to this question is beyond the scope of this paper.

¹¹Future research could explore this issue by explicitly testing whether people care more their reputation of being sincere or they have true intrinsic preferences to keep their words.

(see the Appendix for a detailed description of the design). Here we present the main results from the incentivized survey.

There is a social norm of giving between 40% and 50% of the endowment and a social norm of promise-keeping among politicians.

We find politicians believe a distribution between 40% and 50% of the endowment is socially acceptable in the standard (anonymous) dictator game scenario (see Table 7). In the (anonymous) promise game scenario, promisebreaking is seen as socially unacceptable among politician respondents even promising a lesser amount than a 50:50 split (e.g. between Rs.200 and Rs.400) and keeping that promise is seen as more acceptable than giving the same amount by breaking a promise (i.e. promising Rs.500 and then giving between Rs.200 and Rs.400) (see Table 8).¹² Seventy-five per cent of the politician respondents express that promising a 50:50 split and keeping that promise is socially acceptable (with a mean of 3.51). This suggests that there is a norm of equal-sharing among politicians.

In the incentivized survey, non-politician respondents reveal that (1) equal sharing is socially acceptable (see Table A6); and (2) promising and giving zero is socially unacceptable (see Table A8). However, when asked whether promising an equal-split and giving less than that is socially acceptable or not, on average more politicians believe it is socially unacceptable than do non-politicians, and this difference is statistically significant (Table A8). Given the opportunity to choose a promised amount and actual-giving of a hypothetical dictator, our politicians, on average, chose a promised amount of Rs.565 and a giving of Rs.416 (and the average difference was Rs.147.50). Non-politician respondents, on average, chose a similar promised amount (Rs.570) but their chosen giving was significantly lower than that of politicians (Rs.306). Hence, promise-breaking of the hypothetical dictator perceived by politicians and non-politicians is significantly different (with t-statistic equal to -3.300; see Table A8). These findings suggest that nonpoliticians feel less obligated to keep their promises and behave less generously than do politicians.

¹²The magnitude of the difference between promised amount and the actual giving also matters here. For example, promising 50:50 and then giving zero is seen as 'very socially inappropriate' (with a mean of 1.4) whereas promising 50:50 and giving a little less than that is seen as 'somewhat socially inappropriate' (mean is 2.4). Our politician respondents think that making a zero promise, and keeping that promise, is not socially acceptable too (64% of them express that with a mean value of 1.82).

7 Concluding remarks

Evidence from behavioural economics suggests that an institution that nurtures and upholds social preferences can achieve socially desirable outcomes cost-effectively that might be unattainable by incentives that appeal only to self-interest (Bowles 2008; Cooter 1998; Frey 1997; Ostrom 2000). Herein we investigate whether self-selected politicians' social preferences for keeping their promises and others' well-being help attain fair outcomes in an environment in which it is easy for them to serve their private interests. We show how the combination of two non-monetary routes change politicians' behaviour from zero- to 50:50-giving, bolstering pro-social decisions in this hard-to-govern environment. The first one is a commitment device, a (non-binding) promise, while the second adds a relational lever between the politician-dictator and the recipient.

Despite negative perceptions about politicians and promise-keeping among voters, the political science literature argues that politicians tend to fulfil election pledges. In Thomson et al.'s (2017) study of 57 electoral campaigns in 12 countries, 60-80% of election pledges were kept, with a higher fulfilment for single-party governments than for coalitions. Our lab-in-the-field experiments provide additional and direct evidence of the pro-social effect that non-binding promises and a relational lever have on the behaviour of real politicians. Our paper shows that even in the absence of material incentives induced by repeated interactions with voters and electoral competition and campaigns, it is possible to improve citizens' welfare in the presence of asymmetric information due to the virtue of politicians' social preferences. This suggests that the mechanisms that may stimulate pro-social politician behaviour—even in settings where deciding not to comply with commitments would appear to carry few consequences—are richer than what is typically assumed in the governance literature. Our study suggests that more transparent and informative communication between politicians and their constituencies, not only during electoral campaigns but also when they are in office, could help to provide incentives to politicians to act in favour of their citizens.

One possible concern is whether the promise results (i.e. politicians' promise-keeping and pro-social giving) are caused by an experimenter demand effect (EDE). There are two possible channels through which an EDE could occur. First, politicians could feel under 'special' scrutiny when they received the invitation to participate in the experiment. Second, in the lab a politician-dictator could respond to the explicit presence of the 'audience', including the experimenters (this was a deliberate feature of our design and intended to heighten the social-image effect (as in Andreoni & Bernheim 2009)). On the first point, we are confident that this was not the case, first, because of the emphasis throughout that participation was voluntary. A politician concerned about 'special scrutiny' could simply opt out. Second, and more importantly, our results in T2, where one-third of the dictators gave zero, suggest that scrutiny did not interfere with and discourage selfish behaviour. Since EDEs would be expected to be consistent across the treatments, their limited impacts on behaviour in T2 suggest limited impacts on behaviour in the promise treatment as well. Adding to this, the promised amount could only be observed by the corresponding recipient and never by the experimenters or the other participants. A politician-dictator could also hide his/her distributed amount by choosing zero-giving, which no-one, including the experimenters, would be able to identify. Finally, our supplementary evidence from the incentivized norm elicitation survey helps mitigate the experimenter effect and external validity concerns.

We recognize the following limitations of our study that future research could explore. First, one may question how representative our village-level politicians are. We argue, however, that they become politicians by following a standard election process and exercise substantial power—financial and decision making—in their decentralized everyday setting. Second, our sample size is small. However, it is very challenging to recruit politicians for lab experiments, even at the village level. Future research should investigate politicians' motivations and promise-keeping behaviour with a larger data set and at different levels (e.g. block-/district-level leaders, municipal council leaders) and in different countries.

References

- Akerlof, G. A. (1978). The market for 'lemons': Quality uncertainty and the market mechanism. In Uncertainty in economics, pp. 235–251. New York: Elsevier.
- Andreoni, J. & Bernheim, B. D. (2009). Social image and the 50–50 norm: A theoretical and experimental analysis of audience effects. *Econometrica*, 77(5):1607–1636.

- Anukriti, S. & Chakravarty, A. (2019). Democracy and demography societal effects of fertility limits on local leaders. *Journal of Human Resources*, 54(1):79–121.
- Ariely, D., Bracha, A., & Meier, S. (2009). Doing good or doing well? Image motivation and monetary incentives in behaving prosocially. *American Economic Review*, 99(1):544–555.
- Banuri, S. & Keefer, P. (2016). Pro-social motivation, effort and the call to public service. *European Economic Review*, 83:139–164.
- Bénabou, R. & Tirole, J. (2006). Incentives and prosocial behavior. American Economic Review, 96(5):1652–1678.
- Bernheim, B. D. & Kartik, N. (2014). Candidates, character, and corruption. American Economic Journal: Microeconomics, 6(2):205–246.
- Besley, T. & Coate, S. (1997). An economic model of representative democracy. The Quarterly Journal of Economics, 112(1):85–114.
- Besley, T. & Ghatak, M. (2005). Competition and incentives with motivated agents. American Economic Review, 95(3):616–636.
- Bowles, S. (2008). Policies designed for self-interested citizens may undermine 'the moral sentiments': Evidence from economic experiments. *Science*, 320(5883):1605–1609.
- Buisseret, P. & Prato, C. (2016). Electoral control and the human capital of politicians. *Games and Economic Behavior*, 98:34–55.
- Burger, J. M. (1999). The foot-in-the-door compliance procedure: A multiple-process analysis and review. *Personality and Social Psychology Review*, 3(4):303–325.
- Callander, S. (2008). Political motivations. *The Review of Economic Studies*, 75(3):671–697.
- Callander, S. & Wilkie, S. (2007). Lies, damned lies, and political campaigns. Games and Economic Behavior, 60(2):262–286.

- Calvert, R. L. (1985). Robustness of the multidimensional voting model: Candidate motivations, uncertainty, and convergence. *American Journal* of Political Science, 29:69–95.
- Carlsson, F., Kataria, M., Krupnick, A., Lampi, E., Löfgren, Å., Qin, P., & Sterner, T. (2013). The truth, the whole truth, and nothing but the truth: A multiple country test of an oath script. *Journal of Economic Behavior* & Organization, 89:105–121.
- Charness, G. & Dufwenberg, M. (2006). Promises and partnership. *Econo*metrica, 74(6):1579–1601.
- Charness, G. & Gneezy, U. (2008). What's in a name? Anonymity and social distance in dictator and ultimatum games. *Journal of Economic Behavior* & Organization, 68(1):29–35.
- Chattopadhyay, R. & Duflo, E. (2004). Women as policy makers: Evidence from a randomized policy experiment in India. *Econometrica*, 72(5):1409–1443.
- Cohen T., Wolf, S., Panter, A., & Insko, C. (2011). Introducing the GASP scale: A new measure of guilt and shame proneness. *Journal of Personality* and Social Psychology, 100(5): 947–966.
- Cohen, T., Panter, A., & Turan, N. (2012). Guilt proneness and moral character. Current Directions in Psychological Science, 21(5):355–359.
- Cooter, R. (1998). Expressive law and economics. *The Journal of Legal Studies*, 27(S2):585–607.
- Corazzini, L., Kube, S., & Maréchal, M. A. (2007). Towards behavioral public choice: Guilt aversion and accountability in the lab. ISLA working paper 27.
- Corazzini, L., Kube, S., Maréchal, M. A., & Nicolo, A. (2014). Elections and deceptions: An experimental study on the behavioral effects of democracy. *American Journal of Political Science*, 58(3):579–592.
- Dal Bó, E., Finan, F., Folke, O., Persson, T., & Rickne, J. (2017). Who becomes a politician? The Quarterly Journal of Economics, 132(4):1877– 1914.

- Dana, J., Weber, R. A., & Kuang, J. X. (2007). Exploiting moral wiggle room: Experiments demonstrating an illusory preference for fairness. *Economic Theory*, 33(1):67–80.
- Delfgaauw, J. & Dur, R. (2007). Incentives and workers' motivation in the public sector. *The Economic Journal*, 118(525):171–191.
- Dewan, T. & Shepsle, K. (2011). Political economy models of elections. Annual Review of Political Science, 14:311–330.
- Ellingsen, T. & Johannesson, M. (2004). Promises, threats and fairness. The Economic Journal, 114(495):397–420.
- Enemark, D., Gibson, C. C., McCubbins, M. D., & Seim, B. (2016). Effect of holding office on the behavior of politicians. *Proceedings of the National Academy of Sciences*, 113(48):13690–13695.
- Ferraz, C. & Finan, F. (2011). Electoral accountability and corruption: Evidence from the audits of local governments. *American Economic Review*, 101(4):1274–1311.
- Fisman, R., Schulz, F., & Vig, V. (2014). The private returns to public office. Journal of Political Economy, 122(4):806–862.
- Frey, B. S. (1997). A constitution for knaves crowds out civic virtues. The Economic Journal, 107(443):1043–1053.
- Geng, H., Weiss, A. R., & Wolff, I. (2011). The limited power of voting to limit power. Journal of Public Economic Theory, 13(5):695–719.
- Gneezy, U. (2005). Deception: The role of consequences. American Economic Review, 95(1):384–394.
- Goeree, J. K., McConnell, M. A., Mitchell, T., Tromp, T., & Yariv, L. (2010). The 1/d law of giving. American Economic Journal: Microeconomics, 2(1):183–203.
- Hanna, R. & Wang, S.-Y. (2017). Dishonesty and selection into public service: Evidence from India. American Economic Journal: Economic Policy, 9(3):262–290.

- Hao, L. & Houser, D. (2010). Honest lies. GMA Working Paper in Economics 11–16.
- Holmstrom, B. (1979). Moral hazard and observability. Bell Journal of Economics, 10(1):74–91.
- Jack, B. K. & Recalde, M. P. (2015). Leadership and the voluntary provision of public goods: Field evidence from Bolivia. *Journal of Public Economics*, 122:80–93.
- Jacquemet, N., Joule, R.-V., Luchini, S., & Shogren, J. F. (2013). Preference elicitation under oath. Journal of Environmental Economics and Management, 65(1):110–132.
- Kartik, N. & McAfee, R. P. (2007). Signaling character in electoral competition. American Economic Review, 97(3):852–870.
- Kosfeld, M. & Rustagi, D. (2015). Leader punishment and cooperation in groups: Experimental field evidence from commons management in Ethiopia. American Economic Review, 105(2):747–783.
- Krupka, E. L. & Weber, R. A. (2013). Identifying social norms using coordination games: Why does dictator game sharing vary? Journal of the European Economic Association, 11(3):495–524.
- Leider, S., Möbius, M. M., Rosenblat, T., & Do, Q. A. (2009). Directed altruism and enforced reciprocity in social networks. *The Quarterly Journal* of *Economics*, 124(4): 1815–1851.
- Osborne, M. J. & Slivinski, A. (1996). A model of political competition with citizen-candidates. *The Quarterly Journal of Economics*, 111(1):65–96.
- Ostrom, E. (2000). Crowding out citizenship. Scandinavian Political Studies, 23(1):3–16.
- Pons, V. (2018). Will a five-minute discussion change your mind? A countrywide experiment on voter choice in France. American Economic Review, 108(6): 1322–1363.
- Sastry, T. (2014). Towards decriminalisation of elections and politics. *Economic and Political Weekly*, 49(1):34–41.

- Thomson, R. (2011). Citizens' evaluations of the fulfillment of election pledges: Evidence from Ireland. *The Journal of Politics*, 73(1):187–201.
- Thomson, R., Royed, T., Naurin, E., Artés, J., Costello, R., Ennser-Jedenastik, L., et al. (2017). The fulfillment of parties' election pledges: A comparative study on the impact of power sharing. *American Journal of Political Science*, 61(3):527–542.
- Vanberg, C. (2008). Why do people keep their promises? An experimental test of two explanations 1. *Econometrica*, 76(6):1467–1480.
- Wittman, D. (1983). Candidate motivation: A synthesis of alternative theories. American Political Science Review, 77(1):142–157.

Figures and tables



Figure 1: Kernel density of amount given by politicians in different treatments









Figure 1(e): K-density of give in T2 and T4

Figure 1(f): K-density of give in T3 and T4



Figure 1(d): K-density of give in T2 and T3

Figure 1(b): K-density of give in T1 and T3



Figure 2: Frequency of amount given by politicians across treatments



Figure 3: Kernel density of promise and amount given by politicians in T3 and T4



Figure 3(a): K-density of promise and give in T3

Figure 3(b): K-density of promise and give in T4
Figure 4: Promise and amount given by politicians in T3 and T4



Figure 4(a): Promise and give in T3

Figure 4(b): Promise and give in T4

Table 1: Summary s	tatistics of average	giving, zerc	-giving,	and 50:50-giving	across treat	ments and	pair-wise
comparisons across	treatments (t test)	for politician	is (SD in	parentheses)			

	Average giving (Rs.)	Zero-giving (number of times, %)	50:50-giving (number of times, %)	Average promised amount (Rs.)
T1	13.33 (34.57*)	87	0	-
T2	356.60 (291.88)	27	32	_
Т3	80.33 (86.40)	37	0	87.00 (95.23)
T4	459.67 (302.69)	14	45	480.64 (263.54)
t test				
T1 vs T2	6.39* (0.00)	-6.39* (0.00)	3.71* (0.00)	-
T1 vs T3	3.94* (0.00)	-4.56* (0.00)	-	-
T2 vs T4	1.85*(0.06)	-1.59 (0.11)	1.43 (0.15)	-
T3 vs T4	6.71* (0.00)	-2.46* (0.01)	4.91* (0.00)	7.91* (0.00)
T1 vs T4	8.02* (0.00)	-9.21* (0.00)	4.91* (0.00)	-
T2 vs T3	-5.04* (0.00)	-0.97 (0.33)	3.71* (0.00)	-

Notes: * significant at the 5% level (*p*-values in parentheses). Note that two-sample Wilcoxon rank-sum (Mann–Whitney) tests show similar findings.

Amount given (x)		T1			T2			Т3			T4		
-	#	%	Cum.										
<i>x</i> = 0	26	86.67	86.67	14	26.42	26.42	11	36.67	36.67	9	14.52	14.52	
100 ≥ <i>x</i> > 0	4	13.33	100	1	1.89	28.30	11	36.66	73.33	6	9.68	24.19	
200 ≥ <i>x</i> > 100	-	_	_	3	5.66	33.96	6	20.00	93.33	2	3.23	27.42	
300 ≥ <i>x</i> > 200	-	_	_	9	16.98	50.94	2	6.67	100	1	1.61	29.03	
400 ≥ <i>x</i> > 300	_	_	_	2	3.77	54.72	-	-	_	1	1.61	30.65	
500 <i>> x></i> 400	_	_	_	0	0.0	54.72	-	-	_	0	0	30.65	
<i>x</i> = 500	_	_	_	17	32.08	86.79	-	-	_	28	45.16	75.81	
600 ≥ <i>x</i> >500	_	_	_	1	1.89	88.68	-	-	_	3	4.84	80.65	
700 ≥ <i>x</i> > 600	_	_	_	1	1.89	90.57	-	-	_	1	1.61	82.26	
800 ≥ <i>x</i> >700	_	_	_	0	0.0	90.57	-	-	_	3	4.84	87.10	
900 ≥ <i>x</i> > 800	-	_	_	1	1.89	92.45	_	_	-	1	1.61	88.71	
1000 ≥ <i>x</i> > 900	-	_	_	4	7.55	100.0	_	_	-	7	11.29	100	
Numver of observations		30			53			30			62		

Table 2. Amount given by politicians in four treatments

Amount promised			Т3				T4	
	Amou	nt prom	ised	Amount given – amount promised (mean)	Amc	Amount promised Amount giv – amoun promisec (mean)		
	Number	%	Cum.	Mean	Number	%	Cum.	Mean
0	11	37	37	0	6	9.68	9.68	33.33
10	1	3	40	0	-	-	-	_
50	3	10	50	0	_	_	_	_
100	6	20	70	0	6	9.68	19.35	0
150	5	17	87	0	_	_	_	_
200	1	3	90	0	-	-	-	_
300	3	10	100	-200	1	1.61	20.97	-100
400	-	_	_	-	1	1.61	22.58	0
500	_	_	_	-	33	53.23	75.81	6.06
600	-	_	_	-	4	6.45	82.26	-150
700	_	_	_	-	3	4.84	87.10	-33.33
800	_	_	_	-	2	3.23	90.32	0
900	_	_	_	-	1	1.61	91.94	-900
1000	_	_	_	-	5	8.06	100	0
Total			30				62	

Table 3: Amount promised and mean amount given versus amount promised, politicians

ТЗ				T4			
Give – promise	Number %		Cum. percentage	Number	%	Cum. percentage	
-900	0	0	0	1	1.61	1.61	
-600	0	0	0	1	1.61	3.23	
-500	0	0	0	2	3.23	6.45	
-200	1	3	3	2	3.23	9.68	
-100	0	0	3	1	1.61	11.29	
0	29	97	100	50	80.65	91.94	
100	0	0	100	1	1.61	93.55	
200	0	0	100	1	1.61	95.16	
400	0	0	100	1	1.61	96.77	
500	0	0	100	2	3.23	100	
Number of observations		30			62		

Table 4: Promise-keeping, politicians

Table 5: Regression results (politicians)

	(1)	(2)	(3)
T2	422.91* (std error: 62.77; <i>t</i> - stat: 6.74; <i>p</i> -value: 0.00)	-1.86* (std error: 0.38; <i>z</i> - stat: -4.79; <i>p</i> - value: 0.00)	-
Τ3	62.22 (std error: 66.64; <i>t-</i> stat: 0.93; <i>p</i> -value: 0.35)	-1.25* (std error: 0.39; <i>z</i> - stat: - 3.19; <i>p</i> - value: 0.00)	-
Τ4	448.19* (std error: 56.52; <i>t</i> - stat: 7.93; <i>p</i> -value: 0.00)	-2.11* (std error: 0.36; <i>z</i> - stat: - 5.84; <i>p</i> - value: 0.00)	-
Politician	-	-	0.91* (0.29)
Controls?	Yes	Yes	Yes
Adjusted R ² /Pseudo-R ²	0.36	0.22	0.36
Number of observations	172	166	103

Notes: controls: caste and occupation dummies, years of education, age, gender, state dummy. Column (1) dependent variable: amount given; column (2) dependent variable: dummy if give = 0, 0 otherwise; column (3) dependent variable: dummy if (give – promise) is 0, 0 otherwise. For column (1), the estimator is OLS; for columns (2) and (3), the estimator is probit.

	Give	Adj. R ²	Zero-giving	Pseudo R ²	Controls	Observations
			T1 vs T	2		
T2	367.40* (58.80)	0.61	-1.69* (0.35)	0.27	Yes	83
			T2 vs T	4		
T4	143.48* (55.65)	0.63	-0.37 (0.39)	0.05	Yes	114
			T1 vs T	3		
Т3	61.29* (16.48)	0.50	-1.82* (0.49)	0.41	Yes	60
			T3 vs T	4		
T4	381.04* (56.02)	0.70	-0.73* (0.33)	0.09	Yes	92

Table 6: Summary of regression results, pair-wise comparison of treatments

Table 7: Elicited norm of equal-sharing for politicians

Give	Mean		_	+	++
'Give 0'	1.88 (1.22)	0.60	0.10	0.10	0.20
'Give 100'	2.71 (0.97)	0.13	0.28	0.35	0.24
'Give 200'	2.89 (0.93)	0.11	0.19	0.42	0.28
'Give 300'	2.46 (1.09)	0.26	0.23	0.31	0.20
'Give 400'	3.00 (1.04)	0.14	0.15	0.30	0.41
'Give 500'	2.82 (1.15)	0.18	0.21	0.20	0.41
'Give 600'	2.45 (1.13)	0.28	0.24	0.25	0.23
'Give 700'	1.81 (1.08)	0.56	0.20	0.10	0.14
'Give 800'	2.24 (1.20)	0.40	0.19	0.18	0.23
'Give 900'	2.28 (1.13)	0.31	0.34	0.12	0.23
'Give 1000'	1.91 (1.25)	0.58	0.17	0.01	0.24
Number of observations	120				

Notes: p < 0.05, all one-tailed. Responses are: 'very socially inappropriate' (– –), 'somewhat socially inappropriate' (–), 'somewhat socially appropriate' (+), 'very socially appropriate' (++). To construct the mean ratings, we converted responses into numerical scores ('very socially inappropriate' (– –) = 1, 'somewhat socially inappropriate' (–) = 2, 'somewhat socially appropriate' (+) = 3, 'very socially appropriate' (++) = 4.

Table 8: Elicited norm	of	promise-k	eeping	for	politicians

Action	Mean		_	+	++
Promise Rs.0 and give Rs.0	1.90	63.33	9.17	0.83	26.67
Promise Rs.0 and give more than Rs.0	2.65	13.33	27.50	40.00	19.17
Promise Rs.500 and give Rs.500	3.51	9.17	5.83	9.17	75.83
Promise Rs.500 and give Rs.0	1.42	74.17	14.17	6.67	5.00
Promise Rs.500 and give more than Rs.500	3.12	6.72	16.81	33.61	42.86
Promise Rs.500 and give any amount in between Rs.200 and Rs.400	2.4	23.33	33.33	23.33	20.00
Promise Rs.1000 and give Rs.1000	3.39	15.83	3.33	6.67	74.17
Promise Rs.1000 and give Rs.0	1.39	77.50	10.83	6.67	5.00
Promise Rs.1000 and give Rs.500	2.14	30.00	34.17	27.50	8.33
Promise Rs.1000 and give less than Rs.1000	2.12	31.09	32.77	28.57	7.56
Promise more than Rs.500 and give the promised amount	3.4	9.17	8.33	15.83	66.67
Promise in between Rs.200 and Rs.400 and give the promised amount	3.15	14.29	11.76	18.49	55.46
Promise in between Rs.200 and Rs.400 and give less than the promised amount	2.03	41.67	27.50	16.67	14.17

Notes: --, socially inappropriate; -, somewhat socially inappropriate; +, somewhat socially appropriate; ++, socially appropriate

	Promise- keeping (%)		Averaç (F	ge giving Rs.)	Average (R	e promise Rs.)	50:5	0 (%)	Zero-gi	iving (%)
	Pol.	Non- pol.	Pol.	Non-pol.	Pol.	Non-pol.	Pol.	Non-pol.	Pol.	Non-pol.
T1	x	х	13.33	5	Х	x	0	0	87	93
T2	x	x	356	335	x	x	32.08	35.48	26.42	19.35
ТЗ	96	100	80	41.33	87.33	41.33	0	0	37	43
T4	81	55	460	402	519	467	45.16	35.71	14.52	19.05

Appendix figures and tables



Figure A1: Promise versus amount given (give), non-politicians

Figure A2: Amount given, no promise and promise treatments, non-politicians



Table A1: Summary statistics, politicians and non-politicians characteristics

	Lab-in-the-Field	
	Politicians – Means	Non-Politicians – Means
Female	0.34 (0.47)	0.50 (0.50)
Years of Education	9.64 (3.93)	8.71 (3.88)
Age	40.52 (11.14)	38.01 (12.09)
Forward Caste (per cent)	44%	50%
Other Caste (per cent)	50%	44%
Farmer (per cent)	50%	48%
Number - West Bengal	149	119
Number -Uttar Pradesh	26	14
Total Number	175	133

Note: Standard deviations in brackets.

Table A2: Summary statistics of average giving, zero-giving, and 50:50-giving across treatments and pair wise comparisons across treatments (*ttest*) for non-politicians

	Average giving (Rs)	Zero-giving (%)	50:50-giving (%)	Average promised amount (Rs)
T1	5.00 (20.12)	93	0	
T2	335.48 (256.31)	19	35	
Т3	41.33 (50.63)	43	0	41.33 (9.24)
T4	402.38 (281.53)	19	35	466.66 (230.23)

* SD in parenthesis Ttest: pairwise ttest of average giving, zero-giving, 50:50 giving, and promised amount among non-politicians (two-way ttest, p-values in parenthesis)

among non-politicia	ns (two-way ttest, p-va	alues in parenthesis)		
T1 vs T2	7.03*(0.00)		3.99* (0.00)	
T1 vs T3	3.65* (0.00)	4.85* (0.00)		
T2 vs T4	1.04 (0.30)	0.03 (0.97)	0.02 (0.98)	
T3 vs T4	6.93* (0.00)		4.02* (0.00)	9.92* (0.00)
T1 vs T4	7.70* (0.00)	—9.00* (0.00)	4.02* (0.00)	
T2 vs T3	—6.16* (0.00)	—2.05* (0.04)	3.99* (0.00)	

Amount Given (x)		T1			T2				Т3			T4	
-	#	%	Cum.	#	%	Cum.	_	#	%	Cum.	#	%	Cum.
<i>x</i> = 0	28	93.33	93.33	6	19.35	19.35		13	43.33	43.33	8	19.05	19.05
100 ≥ x> 0	2	6.67	100	4	12.90	32.26		14	46.67	90.00	1	2.28	21.43
200 ≥ x> 100				2	6.45	38.71		3	10.00	100	6	14.29	35.71
300 ≥ x> 200				3	9.68	48.39					1	2.38	38.10
400 ≥ x> 300				2	6.45	54.84					2	4.76	42.86
500 > x> 400				0	0.0	54.84					0	0	42.86
<i>x</i> = 500				11	35.48	90.32					15	35.71	78.57
600 ≥ x>500				1	3.23	93.55					2	4.76	83.33
700 ≥ x> 600				0	0.0	93.55					3	7.14	90.48

800 ≥ x>700		1	3.23	96.77		1	2.38	92.86
900 ≥ x> 800		0	0.0	96.77		1	2.38	95.24
1000 ≥ x> 900		1	3.23	100.0		2	4.76	100
# of Observations	30		31		30		42	

Amount Promised	Amount Promise		Amount Given- Amount Promised		
	Number	%	Cum.	Mean	
0	3	7.14	7.14	0	
100	2	4.76	11.90	400	
200	4	9.52	21.43	0	
300	1	2.38	23.81	0	
400	1	2.38	26.19	-200	
500	21	50.00	76.19	-71.42	
600	3	7.14	83.33	33.33	
700	3	7.14	90.48	-100	
800	2	4.76	95.24	-150	
900	1	2.38	97.62	-300	
1000	1	2.38	100	-1000	
Total Number	42				

Table A4: Amount promised and mean amount given-amount promised, non-politicians

Table A5: Promise-keeping, non-politicians

	T3			T4		
Give-Promise	Number	Per cent	Cum. Percentage	Number	Per cent	Cum. Percentage
-1000	0	0	0	1	2.38	2.38
-500	0	0	0	4	9.52	11.90
-300	0	0	0	3	7.14	19.05
-200	0	0	0	2	4.76	23.81
-100	0	0	0	4	9.52	33.33
0	30	100	100	23	54.76	88.10
100	0	0	100	2	4.76	92.86
500	0	0	100	2	4.76	97.62
800	0	0	100	1	2.38	100
Observation	30			42		

Table A6: Regression results for non-politicians

	(1)	(2)
T2	342.30* (std error: 64.57; t-stat: 5.30; p-value: 0.00)	-1.78* (std error: 0.50; z-stat: - 3.52; p-value: 0.00)
Т3	57.44 (std error: 57.13; t-stat: 1.01; p-value: 0.31)	-1.44* (std error: 0.45; z-stat: - 3.09; p-value: 0.00)
Τ4	407.16* (std error: 61.63; t-stat: 6.61; p-value: 0.00)	-2.02* (std error: 0.35; z-stat: - 4.08; p-value: 0.00)
Controls?	Yes	Yes
Adjusted R-square/Pseudo-R-square	0.46	0.42
No of Observations	131	131

Notes: controls: caste and occupation dummies, years of education, age, gender, state dummy; Col (1): dependent variable: amount given, Col (2): dependent variable: dummy if give=0, 0 otherwise; For col. (1), the estimator is OLS, and for cols. (2), the estimator is probit.

Action	Respondents	Mean		-	+	++	t-statistic
"Give 0"	Politician	1.88 (1.22)	0.60	0.10	0.10	0.20	2.17*
	Non- Politician	1.56 (1.02)	0.72	0.10	0.08	0.10	
"Give 100"	Politician	2.71 (0.97)	0.13	0.28	0.35	0.24	1.88*
	Non- Politician	2.48 (1.02)	0.20	0.30	0.31	0.19	
"Give 200"	Politician	2.89 (0.93)	0.11	0.19	0.42	0.28	1.95*
	Non- Politician	2.66 (0.89)	0.15	0.16	0.55	0.14	
"Give 300"	Politician	2.46 (1.09)	0.26	0.23	0.31	0.20	0.35
	Non- Politician	2.41 (1.07)	0.27	0.21	0.34	0.18	
"Give 400"	Politician	3.00 (1.04)	0.14	0.15	0.30	0.41	0.70
	Non- Politician	2.91 (0.97)	0.12	0.17	0.40	0.31	
"Give 500"	Politician	2.82 (1.15)	0.18	0.21	0.20	0.41	-0.22
	Non- Politician	2.85 (1.15)	0.16	0.24	0.17	0.43	
"Give 600"	Politician	2.45 (1.13)	0.28	0.24	0.25	0.23	0.22
	Non- Politician	2.42 (1.17)	0.30	0.23	0.21	0.26	
"Give 700"	Politician	1.81 (1.08)	0.56	0.20	0.10	0.14	-0.45
	Non- Politician	1.88 (1.17)	0.58	0.11	0.14	0.17	
"Give 800"	Politician	2.24 (1.20)	0.40	0.19	0.18	0.23	0.26
	Non- Politician	2.2 (1.24)	0.44	0.17	0.14	0.25	
"Give 900"	Politician	2.28 (1.13)	0.31	0.34	0.12	0.23	0.65
	Non- Politician	2.18 (1.21)	0.42	0.20	0.14	0.24	
"Give 1000"	Politician	1.91 (1.25)	0.58	0.17	0.01	0.24	0.05
	Non- Politician	1.90 (1.30)	0.63	0.09	0.02	0.26	

Table A7: Elicited norm of equal-sharing for politicians and non-politicians

**p<0.05, all one-tailed

Responses are: "very socially inappropriate" (- -), "somewhat socially inappropriate" (-), "somewhat socially appropriate" (+), "very socially appropriate" (+ +). To construct the mean ratings, we converted responses into numerical scores ("very socially inappropriate" (- -) = 1, "somewhat socially inappropriate" (-) = 2, "somewhat socially appropriate" (+ +) = 3, "very socially appropriate" (+ +) = 4).

Table A8: Elicited norm of promise-keeping for politicians and non-politicians

Action	Respondents	Mean		-	+	++	t-statistic
Promise Rs. 0 and Give Rs 0	Politician	1.90	63.33	9.17	0.83	26.67	0.60
	Non-Politician	1.80	66.67	9.17	0.83	23.33	
Promise Rs. 0 and Give more than Rs 0	Politician	2.65	13.33	27.50	40.00	19.17	-1.81*
	Non-Politician	2.85	6.67	22.50	49.17	21.67	
Promise Rs 500 and Give Rs 500	Politician	3.51	9.17	5.83	9.17	75.83	0.49
	Non-Politician	3.45	6.67	5.83	22.50	65.00	
Promise Rs 500 and Give Rs 0	Politician	1.42	74.17	14.17	6.67	5.00	0.15
	Non-Politician	1.40	77.50	9.17	8.33	5.00	
Promise Rs 500 and more than Rs 500	Politician	3.12	6.72	16.81	33.61	42.86	2.00*
	Non-Politician	2.86	14.41	22.03	26.27	37.29	
Promise Rs 500 and give any amount in between Rs 200 and Rs 400	Politician	2.4	23.33	33.33	23.33	20.00	-3.30*
	Non-Politician	2.85	13.33	23.33	28.33	35.00	
Promise Rs. 1000 and Give Rs 1000	Politician	3.39	15.83	3.33	6.67	74.17	0.89
	Non-Politician	3.25	17.50	6.67	8.33	67.50	
Promise Rs. 1000 and Give Rs 0	Politician	1.39	77.50	10.83	6.67	5.00	0.72
	Non-Politician	1.31	82.50	8.33	4.17	5.00	
Promise Rs 1000 and Give Rs 500	Politician	2.14	30.00	34.17	27.50	8.33	-2.31*
	Non-Politician	2.42	19.17	32.50	35.00	13.33	
Promise Rs 1000 and Give less than Rs 1000	Politician	2.12	31.09	32.77	28.57	7.56	-2.63*
	Non-Politician	2.46	25.00	20.83	36.67	17.50	
Promise more than Rs 500 and give the promised amount	Politician	3.4	9.17	8.33	15.83	66.67	0.04
	Non-Politician	3.39	6.72	9.24	21.85	62.18	
Promise in between Rs 200 and Rs 400 and give the promised amount	Politician	3.15	14.29	11.76	18.49	55.46	-0.06
	Non-Politician	3.15	9.24	17.65	21.01	52.10	
Promise in between Rs 200 and Rs 400 and give less than the promised amount	Politician	2.03	41.67	27.50	16.67	14.17	-1.76*
	Non-Politician	2.29	34.17	27.50	13.33	25.00	

(--) => Socially Inappropriate; (-) => Somewhat Socially Inappropriate;; (+) => Somewhat Socially Appropriate; (++) => Socially Appropriate. Standard errors are in parenthesis. Number of observation is 240.

Appendix

Experimental design

Our implementation of the design is described in the following steps.

No promise, anonymous (i.e., NPA) treatment (T1)

(1) Participants from the home-village (where the venue was located) and the visitorvillage (i.e., from distant locations)--10 politicians and 10 non-politicians from each village-arrived separately at the experimental venue and seated in two different rooms. Participants from home-village did not meet participants from visitor-village before entering the lab, during the experiment, or after the experiment. In each room, there were 20 participants--10 politicians and 10 common villagers--and they were from the same village. (3) The experimenter read out and explained the instructions of the game aloud and answered questions from participants. Each participant was then asked to solve a short quiz. Those who could not answer the quiz properly were given an extra explanation from the experimenter. The experimenter made clear that participant names would not be recorded. No communication between subjects was allowed (verbal or any other type). Two practicerounds of the game were played. (4) Participants in each room were divided into two sections (e.g., A and B) randomly. Each participant in each section in each room was randomly matched with another participant from the same section sitting in other room (e.g., participant x in Section A in Room 1 was randomly matched with participant y from Section A in Room 2) and formed a pair. (5) Participants were told that they would never meet their partners. (6) In each pair, their roles in the game (dictator (D) or recipient (R)) were determined randomly and both politicians and non-politicians could be assigned the role of dictator. We did not change their roles in each round---a randomly chosen dictator remained dictator for the entire session. (7) Each pair received a fixed and known endowment---1000 INR (approx 15.50 $(USD)^{1}$ ---for each round and the dictator had to decide how to allocate the endowment between him/herself and his/her partner (recipient, R) sitting in the next room.

(8) Each D received a random (and confidential) private number between 1 and 10---no other person in the room, not even the experimenter--would know this number. Each D was asked to come in the front desk one by one where each of them picked a chit randomly from an urn containing 10 chits and each chit had a number between 1 and 10. Only a D could see his/her private number, on one else (not even the experimenter). They wrote the number on their decision sheet in private. (9) At the start of each round, the experimenter announced two numbers randomly chosen between 1 and 10 and only those Ds with the corresponding numbers made a decision, other Ds could not. (10) Each D received a decision sheet. They filled in their decision sheets (e.g., their group number, private number, and round number) in an enclosed area one by one. Only Ds whose private numbers were announced could choose and record a distribution on the decision sheet in private, others would just tick a box which

¹ As our research focus was on politician behaviour and it was challenging to recruit politician participants, we did not follow an equal split while randomly allocating the roles of politicians and non-politicians. Instead, we did the following for each session: randomly chose a number between 5 and 8 and chose the split accordingly (e.g., if number 6 is drawn, 6 out 10 politicians played the role of dictator for that session).

stated the nature would give zero to either D or R (see the example of decision sheet below). (11) All Ds, who made a decision or who ticked a box, folded the decision sheets and put them in an envelope, named, e.g., Round 1-Decisions (Section A or B), themselves. No one (including the experimenter) should be able to identify, during and after the experiment, which individual made a decision and what his or her decision was. The Rs sitting next room, other Ds, and the experimenter knew the probability (i.e., 0.8), but did not know whether nature or D made the decision when the outcome was either zero or the entire endowment of 1000 INR (this can only be true if D chooses the same division as nature).

(12) We repeated steps (9) to (11) four times more and each time the experimenter announced different private numbers (i.e., they played 5 rounds). (13) At the end, one of the five rounds was selected randomly to determine the payments. (14) The envelope of decision sheets for that round was given to an external person waiting outside the venue. The external had no information about the game or about the participants. He observed the decision-sheets of different dictators in a separate room and put the payment in a separate envelope for each dictator and recipient. He also decided whether D or R got INR 1000 when nature intervened by flipping a coin. (15) No one in any of the rooms would know D's actual decision, not even the experimenter. (16) Meanwhile, participants filled in a short questionnaire that covered education, occupation and other demographic and related questions. (17) The external person gave a result sheet (see below) for each Section to the experimenter who then showed the results to each Ds of a Section (A or B) and their corresponding Rs of the same Section sitting in the other room. (18) Each participant left the room one-by-one and received their envelopes from the external with their payments (their earning from the game plus a fixed participation fee of Rs. 300), based on the decisions they or their partners or the nature made, outside the room in a separate enclosed area and left the venue one by one. Participants from the visitor village left the venue first.

No-promise, non-anonymous (i.e., NPNA) treatment (T2)

We followed the following steps. (i) Participants from the home-village and the visitorvillage--10 politicians and 10 non-politicians from each village--arrived separately at the experimental venue. Out of 20 participants from the visitor village, 10 participants were randomly chosen to be seated in one room and the others seated in the next room (e.g., each participant from the village randomly picked a chit which showed their room and seat numbers). Similar procedure was followed for home-village participants. Participants from these two different locations did not meet each other before entering the lab. (ii) Following a random matching protocol, a subject from the visitor-village formed a pair/group with a subject from the home-village in each session. (iii) Same as (3) in T1. (iv) Pair members were asked to stand up and greet each other. This was done to increase the moral costs of selfish behaviour (as in AB 2009). (v) Then we followed steps (6), (7), and (8) as in T1. (vi) The Steps (9), (10) and (11) as in T1². (vii) We repeated Step (vi) four times more (i.e., five rounds were played) and each time the experimenter announced different private numbers.

² The Ds who did not make decisions were asked to put a tick in the decision sheet. This was to ensure that recipients would not know that nature made the decision in these cases.

(viii) See (13), (14), and (16) in T1. (ix) The experimenters received the result and envelopes with cash payments for each subject (each subject's individual id number, same as their seat numbers, written on each envelope) from the external. The experimenters published the result (wrote each pair's earning on a board). Note that if a D chose to give 0 (or 1000), no one in the room could identify whether the D or nature made the decision. But, for any other chosen amount, everybody could understand that the D chose the amount. (x) Each participant received their envelopes with payments and left the venue one by one. The subjects from the visitor-village exited before the local participants .

Promise, anonymous (i.e., PA) Treatment (T3)

We followed step (1) to (8) as in T1. Then all Ds wrote (their pair number, not their private number, and) how they would allocate INR 1000 between him/herself and the R on a `Promise Slip'. Each D went to an enclosed area and wrote this in private: the D then put the folded the promise-slip into an enclosed envelope and returned it to the experimenter. The experimenter then carried the promise-slip and passed it (without seeing it) on to the respective R sitting next room. Each R observed what his/her partner promised to give him/her in private. The slip was then folded and put in an envelope called Round 1 which is a general envelope specific to a Round. No one in the room except the respective R, not even the experimenter, could observe the promise. The dictator game described in T1 (i.e., Step (9) to (11)) was then played. After that, we repeated the promise-making stage, as described above, and then Step (9) to (11) again four times more and for a total of five rounds. Steps (13) to (18) were then followed.

Promise, non-anonymous (i.e., PNA) Treatment (T4)

We followed step (i) to (v) as in T2. Then all Ds wrote (their pair number, not their private number, and round) how they would allocate INR 1000 between him/herself and the R on a `Promise Slip'. Each D went to an enclosed area and wrote this in private: the D then folded the promise-slip and returned it to the experimenter who passed it (without seeing it) on to the respective R. Each R observed what his/her partner promised to give him/her in private. The slip was then folded and put in an envelope called Round 1 which is a general envelope specific to a Round. No one in the room except the respective R, not even the experimenter, could observe the promise. The dictator game described in T2 (i.e., Step (vi) to (vii)) was then played. After that, we repeated the promise-making stage, as described above, and then Step (vi) to (vii) again four times more. Steps (viii) to (x) were then followed.

It is important to emphasise that (i) a D can hide his/her actual decision about allocation with a probability of 0.8 (i.e, in each round 2 of 10 dictators would make allocation decisions) and that this probability is fixed and common knowledge; (ii) a D's promise is only seen by the respective R. If a D wants keep everything for her/himself without losing his/her image in front of the respective R, he could simply make a generous promise to R and then give 0. Then nobody---except the D in question---would know whether nature intervened or the D decided. Accordingly and to reiterate , the D can behave selfishly without being `found out' by the `audience'. Notice that any other distribution would reveal the identity of the D who made the decision.

Instructions (Treatment 1) GROUP NUMBER: _____

Welcome

Welcome. Thank you for taking the time to come today. [Introduce PB, SM, and the assistants.] You can ask any of us questions during today's programme.

Thank you again for participating. Just for agreeing to participate you will automatically be given Rs 300 as a "thank you" payment. Anything else you earn today will be in addition to this.

Your Group Number

Your name will never be recorded in this study, or revealed to anyone. In particular, the researchers will not have any access to your identities (if you have given your names/address etc to your local surveyor, this information will remain with them). Instead, you will be known by your Group Number. You have drawn this number randomly when you have entered the room (also, the number is shown above).

Your Partner

You will be paired with another person sitting in different room in this venue. We'll call this person your *partner*. In your group number, the first letter indicates your Section (A or B) and next number indicates your Room (1 or 2). Suppose your group number is A1x. You will be matched with A2x, that is someone who is in Section A in Room 2 with a number x. You will never meet (or know the identity) of your partner. Your partner sitting in a different will meet you (or know your identity). The decisions made today will concern how much money you and your partner earn.

Your Task

Your group has been given Rs 1000 to divide between the two of you. Although you and your partner are in the same group, only one of the two partners will have responsibility for deciding for how to divide the Rs 1000.

Even though only one of you makes decisions, it is very important for everyone to understand how decisions will be made, so please pay attention to all of the instructions.

Decision making partner

We will randomly select one person from each pair as Decision Making partner (or D). The decision making partner could be selected from this room or from the other room. We will ask each person on the right/left side of the room to pick a chit from an urn. In each chit, it is written either D or R (recipient partner). If you pick D, then you will make decision how to allocate Rs 100 between you and your partner. We will explain below the procedure of choosing your decision.

Private number:

Each D should please come in the front desk one by one. Each of you will pick a chit randomly from an urn containing 10 chits and each chit has a number between 1 and 10. We call it as 'private number'—it is private and confidential. Only a D can see his/her private number, on one else (not even the experimenter). Please write the number on your decision sheet in private and do not show this to anybody in the room.

Then one of the following two things will happen:

EITHER...

We'll let the decision making partner chose a division of the Rs 1000 by filling in a line like the following (see the example of Decision Sheet):

"Distribute Rs 1000: I allocate ______ to myself, and ______ to my partner."

Notice that the amounts in the two blank spaces must sum to Rs 1000.

No one here will see what this person writes – not even his/her partner.

In 2 out of 10 cases, the decision making partner will choose.

OR...

We will automatically allocate Rs 1000 to one partner and Rs 0 to the other partner. Someone in another room will flip a coin to determine which partner gets Rs 1000 and which get Rs 0.

In 8 out of 10 cases, the decision making partner will be forced to choose.

We will announce at the beginning of each round who can make a decision. For example, we may say those who have private numbers 3 and 8 will choose the distribution. Others, will tick the box where it shows he/she has no control over decision making.

Everyone in this room will know how the Rs 1000 was divided between the two partners in each group. But no one will be able to understand, even the experimenter, whether the decision making partner made this choice, or whether nature made it automatically. No one will be able to understand what private number the deciding partner received, not even the experimenter, or whether the coin flip came up heads or tails.

Thinking about this from the point of view of the decision maker:

- If your division is Rs 1000 for yourself and Rs 0 for your partner, no one will know whether this was your choice, or our choice.
- Likewise, if your division is Rs 0 for yourself and Rs 1000 for your partner, no one will know whether this was your choice, or our choice.
- However, if you choose any other division say Rs 50, Rs 200, Rs 500 or Rs 700 for yourself and the rest for your partner everyone will be able to figure out that you are responsible for this choice.

Thinking about this from the point of view of the other partner:

- If you are allocated Rs 0, you won't know whether your partner made this choice, or whether we made it.
- Likewise, if you are allocated Rs 1000, you won't know whether your partner made this choice, or whether we made it
- However, if you are allocated any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that your partner is responsible for this choice.

Thinking about this from the point of view of everyone else (including the experimenter) in the room:

- If you see that a decision maker is allocated Rs 0, you won't know whether he/she made this choice, or whether we made it.
- Likewise, if you see that a decision maker is allocated Rs 1000, you won't know whether he/she made this choice, or whether we made it.
- However, if any partner receives any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that the decision making partner is responsible for this choice.

The Decision Sheets

The decision maker will actually receive five sheets, and need to make five different decisions. All of the decisions have the same form as the one we've just described. See the example below.

Only one of these decisions will count. After all decisions are made we will randomly select one of the five decision sheets and use only that one decision sheet to determine payments. It makes good sense, therefore, to make each decision as if it will actually be carried out.

We're going to start the random draw of private number. One by one, each decision maker will come to the front of the room, carrying the envelope containing the blank decision sheets. There he/will pick a chit and a number from 1 to 10 comes up. The number on the chit will be his *private number*. To make sure he doesn't forget this number, he'll write it on each decision sheet before returning to his station. *No one else will see this number*.

At the end, we will publish the result as described above to each participant of each member, i.e., all the decision making partners in Section A of this room and all the recipients in Section A in other room will see the Results Sheet (see an example (see 'Result Sheet') below).

Instructions (Treatment 2) GROUP NUMBER: _____

Welcome

Welcome. Thank you for taking the time to come today. [Introduce PB, SM, and the assistants.] You can ask any of us questions during today's programme.

Thank you again for participating. Just for agreeing to participate you will automatically be given Rs 300 as a "thank you" payment. Anything else you earn today will be in addition to this.

Your Group Number

Your name will never be recorded in this study, or revealed to anyone. In particular, the researchers will not have any access to your identities (if you have given your names/address etc to your local surveyor, this information will remain with them). Instead, you will be known by your Group Number. You have drawn this number randomly when you have entered the room (also, the number is shown above).

Your Partner

You will be paired with another person in the room today. We'll call this person your *partner*. The decisions made today will concern how much money you and your partner earn.

Before we tell you about the decisions, we will take a minute to introduce you to your partner. You and your partner have the same Group Number, but are sitting on opposite sides of the room.

We'll start at the front of the room. We will first ask the two in Group Number 1 to stand and face each other. Then each should say to their partner, "Hello (or wave their hands to each other)." We'll then ask Group 2 to do the same, and will repeat this for all groups.

Your Task

Your group has been given Rs 1000 to divide between the two of you. Although you and your partner are in the same group, only one of the two partners will have responsibility for deciding for how to divide the Rs 1000.

Even though only one of you makes decisions, it is very important for everyone to understand how decisions will be made, so please pay attention to all of the instructions.

Decision making partner

We will randomly select one person from each pair as Decision Making partner (or D). We will ask each person on the right/left side of the room to pick a chit from an urn. In each chit, it is written either D or R (recipient partner). If you pick D, they you will make decision how to allocate Rs 100 between you and your partner. We will explain below the procedure of choosing your decision.

Private number:

Each D should please come in the front desk one by one. Each of you will pick a chit randomly from an urn containing 10 chits and each chit has a number between 1 and 10. We

call it as 'private number'—it is private and confidential. Only a D can see his/her private number, on one else (not even the experimenter). Please write the number on your decision sheet in private and do not show this to anybody in the room.

Then one of the following two things will happen:

EITHER...

We'll let the decision making partner chose a division of the Rs 1000 by filling in a line like the following (see the example of Decision Sheet):

"Distribute Rs 1000: I allocate ______ to myself, and ______ to my partner."

Notice that the amounts in the two blank spaces must sum to Rs 1000.

No one here will see what this person writes – not even his/her partner.

In 2 out of 10 cases, the decision making partner will choose.

OR...

We will automatically allocate Rs 1000 to one partner and Rs 0 to the other partner. Someone in another room will flip a coin to determine which partner gets Rs 1000 and which get Rs 0.

In 8 out of 10 cases, the decision making partner will be forced to choose.

We will announce at the beginning of each round who can make a decision. For example, we may say those who have private numbers 3 and 8 will choose the distribution. Others, will tick the box where it shows he/she has no control over decision making.

Everyone in this room will know how the Rs 1000 was divided between the two partners in each group. But no one will be able to understand, even the experimenter, whether the decision making partner made this choice, or whether nature made it automatically. No one will be able to understand what private number the deciding partner received, not even the experimenter, or whether the coin flip came up heads or tails.

Thinking about this from the point of view of the decision maker:

- If your division is Rs 1000 for yourself and Rs 0 for your partner, no one will know whether this was your choice, or our choice.
- Likewise, if your division is Rs 0 for yourself and Rs 1000 for your partner, no one will know whether this was your choice, or our choice.
- However, if you choose any other division say Rs 50, Rs 200, Rs 500 or Rs 700 for yourself and the rest for your partner everyone will be able to figure out that you are responsible for this choice.

Thinking about this from the point of view of the other partner:

• If you are allocated Rs 0, you won't know whether your partner made this choice, or whether we made it.

- Likewise, if you are allocated Rs 1000, you won't know whether your partner made this choice, or whether we made it
- However, if you are allocated any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that your partner is responsible for this choice.

Thinking about this from the point of view of everyone else (including the experimenter) in the room:

- If you see that a decision maker is allocated Rs 0, you won't know whether he/she made this choice, or whether we made it.
- Likewise, if you see that a decision maker is allocated Rs 1000, you won't know whether he/she made this choice, or whether we made it.
- However, if any partner receives any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that the decision making partner is responsible for this choice.

The Decision Sheets

The decision maker will actually receive five sheets, and need to make five different decisions. All of the decisions have the same form as the one we've just described. See the example below.

Only one of these decisions will count. After all decisions are made we will randomly select one of the five decision sheets and use only that one decision sheet to determine payments. It makes good sense, therefore, to make each decision as if it will actually be carried out.

We're going to start the random draw of private number. One by one, each decision maker will come to the front of the room, carrying the envelope containing the blank decision sheets. There he/will pick a chit and a number from 1 to 10 comes up. The number on the chit will be his *private number*. To make sure he doesn't forget this number, he'll write it on each decision sheet before returning to his station. *No one else will see this number*.

At the end, we will publish the result as described above. See an example (see 'Result Sheet') below how we publish the results.

Instructions (Treatment 3) GROUP NUMBER: _____

Welcome

Welcome. Thank you for taking the time to come today. [Introduce PB, SM, and the assistants.] You can ask any of us questions during today's programme.

Thank you again for participating. Just for agreeing to participate you will automatically be given Rs 300 as a "thank you" payment. Anything else you earn today will be in addition to this.

Your Group Number

Your name will never be recorded in this study, or revealed to anyone. In particular, the researchers will not have any access to your identities (if you have given your names/address etc to your local surveyor, this information will remain with them). Instead, you will be known by your Group Number. You have drawn this number randomly when you have entered the room (also, the number is shown above).

Your Partner

You will be paired with another person sitting in different room in this venue. We'll call this person your *partner*. In your group number, the first letter indicates your Section (A or B) and next number indicates your Room (1 or 2). Suppose your group number is A1x. You will be matched with A2x, that is someone who is in Section A in Room 2 with a number x. You will never meet (or know the identity) of your partner. Your partner sitting in a different will meet you (or know your identity). The decisions made today will concern how much money you and your partner earn.

Your Task

Your group has been given Rs 1000 to divide between the two of you. Although you and your partner are in the same group, only one of the two partners will have responsibility for deciding for how to divide the Rs 1000.

Even though only one of you makes decisions, it is very important for everyone to understand how decisions will be made, so please pay attention to all of the instructions.

Decision making partner

We will randomly select one person from each pair as Decision Making partner (or D). The decision making partner could be selected from this room or from the other room. We will ask each person on the right/left side of the room to pick a chit from an urn. In each chit, it is written either D or R (recipient partner). If you pick D, then you will make decision how to allocate Rs 100 between you and your partner. We will explain below the procedure of choosing your decision.

Private number:

Each D should please come in the front desk one by one. Each of you will pick a chit randomly from an urn containing 10 chits and each chit has a number between 1 and 10. We call it as 'private number'—it is private and confidential. Only a D can see his/her private number, on one else (not even the experimenter). Please write the number on your decision sheet in private and do not show this to anybody in the room.

Then one of the following two things will happen:

EITHER...

We'll let the decision making partner chose a division of the Rs 1000 by filling in a line like the following (see the example of Decision Sheet):

"Distribute Rs 1000: I allocate ______ to myself, and ______ to my partner."

Notice that the amounts in the two blank spaces must sum to Rs 1000.

No one here will see what this person writes – not even his/her partner.

In 2 out of 10 cases, the decision making partner will choose.

OR...

We will automatically allocate Rs 1000 to one partner and Rs 0 to the other partner. Someone in another room will flip a coin to determine which partner gets Rs 1000 and which get Rs 0.

In 8 out of 10 cases, the decision making partner will be forced to choose.

We will announce at the beginning of each round who can make a decision. For example, we may say those who have private numbers 3 and 8 will choose the distribution. Others, will tick the box where it shows he/she has no control over decision making.

Everyone in this room will know how the Rs 1000 was divided between the two partners in each group. But no one will be able to understand, even the experimenter, whether the decision making partner made this choice, or whether nature made it automatically. No one will be able to understand what private number the deciding partner received, not even the experimenter, or whether the coin flip came up heads or tails.

Thinking about this from the point of view of the decision maker:

- If your division is Rs 1000 for yourself and Rs 0 for your partner, no one will know whether this was your choice, or our choice.
- Likewise, if your division is Rs 0 for yourself and Rs 1000 for your partner, no one will know whether this was your choice, or our choice.
- However, if you choose any other division say Rs 50, Rs 200, Rs 500 or Rs 700 for yourself and the rest for your partner everyone will be able to figure out that you are responsible for this choice.

Thinking about this from the point of view of the other partner:

- If you are allocated Rs 0, you won't know whether your partner made this choice, or whether we made it.
- Likewise, if you are allocated Rs 1000, you won't know whether your partner made this choice, or whether we made it

• However, if you are allocated any other amount – say Rs 50, Rs 200, Rs 500 or Rs 700 – you'll know that your partner is responsible for this choice.

Thinking about this from the point of view of everyone else (including the experimenter) in the room:

- If you see that a decision maker is allocated Rs 0, you won't know whether he/she made this choice, or whether we made it.
- Likewise, if you see that a decision maker is allocated Rs 1000, you won't know whether he/she made this choice, or whether we made it.
- However, if any partner receives any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that the decision making partner is responsible for this choice.

Promise making

Before the experimenter announces two private numbers who can make decision, the decision making partner will make a promise. He will write in a promise slip how he/she will want to distribute the Rs 1000 between him/herself and his/her partner (see the example of a Promise Slip) sitting in other room.

Then he/she will fold it and give it to the experimenter and the experimenter will carry the slip to his/her partner in the group who is sitting in the other room. After seeing the promise, the receiver will fold the slip and give it back to the experimenter. No one else, not even the experimenter, will see this.

Then the experimenter will announce which two private numbers can choose an allocation and the decision making partner will actually decide how to distribute the money. We will follow the procedure described above.

You will play five rounds and each round, before knowing whether you can choose a decision or not, you will fill in the Promise Slip as described above.

The Decision Sheets

The decision maker will actually receive five sheets, and need to make five different decisions. All of the decisions have the same form as the one we've just described. See the example below.

Only one of these decisions will count. After all decisions are made we will randomly select one of the five decision sheets and use only that one decision sheet to determine payments. It makes good sense, therefore, to make each decision as if it will actually be carried out.

We're going to start the random draw of private number. One by one, each decision maker will come to the front of the room, carrying the envelope containing the blank decision sheets. There he/will pick a chit and a number from 1 to 10 comes up. The number on the chit will be his *private number*. To make sure he doesn't forget this number, he'll write it on each decision sheet before returning to his station. *No one else will see this number*.

At the end, we will publish the result as described above to each participant of each member, i.e., all the decision making partners in Section A of this room and all the recipients in Section A in other room will see the Results Sheet (see an example (see 'Result Sheet') below).

Instructions (Treatment 4) GROUP NUMBER: _____

Welcome

Welcome. Thank you for taking the time to come today. [Introduce PB, SM, and the assistants.] You can ask any of us questions during today's programme.

Thank you again for participating. Just for agreeing to participate you will automatically be given Rs 300 as a "thank you" payment. Anything else you earn today will be in addition to this.

Your Group Number

Your name will never be recorded in this study, or revealed to anyone. In particular, the researchers will not have any access to your identities (if you have given your names/address etc to your local surveyor, this information will remain with them). Instead, you will be known by your Group Number. You have drawn this number randomly when you have entered the room (also, the number is shown above).

Your Partner

You will be paired with another person in the room today. We'll call this person your *partner*. The decisions made today will concern how much money you and your partner earn.

Before we tell you about the decisions, we will take a minute to introduce you to your partner. You and your partner have the same Group Number, but are sitting on opposite sides of the room.

We'll start at the front of the room. We will first ask the two in Group Number 1 to stand and face each other. Then each should say to their partner, "Hello (or wave their hands to each other)." We'll then ask Group 2 to do the same, and will repeat this for all groups.

Your Task

Your group has been given Rs 1000 to divide between the two of you. Although you and your partner are in the same group, only one of the two partners will have responsibility for deciding for how to divide the Rs 1000.

Even though only one of you makes decisions, it is very important for everyone to understand how decisions will be made, so please pay attention to all of the instructions.

Decision making partner

We will randomly select one person from each pair as Decision Making partner (or D). We will ask each person on the right/left side of the room to pick a chit from an urn. In each chit, it is written either D or R (recipient partner). If you pick D, they you will make decision how to allocate Rs 100 between you and your partner. We will explain below the procedure of choosing your decision.

Private number:

Each D should please come in the front desk one by one. Each of you will pick a chit randomly from an urn containing 10 chits and each chit has a number between 1 and 10. We call it as 'private number'—it is private and confidential. Only a D can see his/her private

number, on one else (not even the experimenter). Please write the number on your decision sheet in private and do not show this to anybody in the room.

Then one of the following two things will happen:

EITHER...

We'll let the decision making partner chose a division of the Rs 1000 by filling in a line like the following (see the example of Decision Sheet):

"Distribute Rs 1000: I allocate ______ to myself, and _____ to my partner."

Notice that the amounts in the two blank spaces must sum to Rs 1000.

No one here will see what this person writes – not even his/her partner.

In 2 out of 10 cases, the decision making partner will choose.

OR...

We will automatically allocate Rs 1000 to one partner and Rs 0 to the other partner. Someone in another room will flip a coin to determine which partner gets Rs 1000 and which get Rs 0.

In 8 out of 10 cases, the decision making partner will be forced to choose.

We will announce at the beginning of each round who can make a decision. For example, we may say those who have private numbers 3 and 8 will choose the distribution. Others, will tick the box where it shows he/she has no control over decision making.

Everyone in this room will know how the Rs 1000 was divided between the two partners in each group. But no one will be able to understand, even the experimenter, whether the decision making partner made this choice, or whether nature made it automatically. No one will be able to understand what private number the deciding partner received, not even the experimenter, or whether the coin flip came up heads or tails.

Thinking about this from the point of view of the decision maker:

- If your division is Rs 1000 for yourself and Rs 0 for your partner, no one will know whether this was your choice, or our choice.
- Likewise, if your division is Rs 0 for yourself and Rs 1000 for your partner, no one will know whether this was your choice, or our choice.
- However, if you choose any other division say Rs 50, Rs 200, Rs 500 or Rs 700 for yourself and the rest for your partner everyone will be able to figure out that you are responsible for this choice.

Thinking about this from the point of view of the other partner:

• If you are allocated Rs 0, you won't know whether your partner made this choice, or whether we made it.

- Likewise, if you are allocated Rs 1000, you won't know whether your partner made this choice, or whether we made it
- However, if you are allocated any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that your partner is responsible for this choice.

Thinking about this from the point of view of everyone else (including the experimenter) in the room:

- If you see that a decision maker is allocated Rs 0, you won't know whether he/she made this choice, or whether we made it.
- Likewise, if you see that a decision maker is allocated Rs 1000, you won't know whether he/she made this choice, or whether we made it.
- However, if any partner receives any other amount say Rs 50, Rs 200, Rs 500 or Rs 700 you'll know that the decision making partner is responsible for this choice.

Promise making

Before the experimenter announces two private numbers who can make decision, the decision making partner will make a promise. He will write in a promise slip how he/she will want to distribute the Rs 1000 between him/herself and his/her partner (see the example of a Promise Slip).

Then he/she will fold it and give it to the experimenter and the experimenter will deliver the slip to his/her partner in the group. After seeing the promise, the receiver will fold the slip and give it back to the experimenter. No one else, not even the experimenter, will see this.

Then the decision making partner will actually decide how to distribute the money. Here's the basic procedure you'll use to distribute the Rs 1000.

You will play five rounds and each round, before knowing whether you can choose a decision or not, you will fill in the Promise Slip as described above.

The Decision Sheets

The decision maker will actually receive five sheets, and need to make five different decisions (and of course five Promise Slip). All of the decisions have the same form as the one we've just described. See the example below.

Only one of these decisions will count. After all decisions are made we will randomly select one of the five decision sheets and use only that one decision sheet to determine payments. It makes good sense, therefore, to make each decision as if it will actually be carried out.

We're going to start the random draw of private number. One by one, each decision maker will come to the front of the room, carrying the envelope containing the blank decision sheets. There he/will pick a chit and a number from 1 to 10 comes up. The number on the chit will be his *private number*. To make sure he doesn't forget this number, he'll write it on each decision sheet before returning to his station. *No one else will see this number*.

At the end, we will publish the result as described above. See an example (see 'Result Sheet') below how we publish the results.

Promise Slip

My group number is _____ Round _____

If I get the opportunity to decide, I will divide the group's fund (i.e, Rs 1000) between myself and my partner in the following way:



To me



Rs _____

To my partner



Rs _____

Decision Sheet

How do you distribute Rupees 1000 ?

A or B

My group number is _____ My private number is _____ Round____





To me



To my partner





In 2 out of 10 cases, dictator has made the decision In 8 out of 10 cases, the decision has been forced

RESULT SHEET

Chosen Decision Sheet: _____

Who made the allocation:

In _____ out of 10 cases, dictator has made the decision

In _____out of 10 cases, the decision has been forced

Group 1 Decision maker:	Rs;	Partner: Rs
Group 2 Decision maker:	Rs;	Partner: Rs
Group 3 Decision maker:	Rs;	Partner: Rs.
Group 4 Decision maker:	Rs;	Partner: Rs.
Group 5 Decision maker:	Rs;	Partner: Rs.
Group 6 Decision maker:	Rs;	Partner: Rs.
Group 7 Decision maker:	Rs;	Partner: Rs.
Group 8 Decision maker:	Rs;	Partner: Rs.
Group 9 Decision maker:	Rs;	Partner: Rs.
Group 10 Decision maker	Rs	; Partner: Rs





Invitation Letter

Name..... Respected Sir/ Madam, Address-----

Indian Statistical Institute (ISI) and Manchester University (MU) in order to understand the perception of people at various layers of society regarding rural development has selected some villages in the state of West Bengal based on a public lottery. To facilitate such a research study, meetings have been planned at the nearest Community Hall or Gram Panchayat office (GP) of the respective villages very soon. We are happy to inform you that your name has been selected based on a random draw. Your participation in this study is completely voluntary at any stage and would be deeply appreciated.

The villagers who agree to participate in the study would receive a token honorarium of Rs 300 as a mark of gratitude. In addition, there is a scope to earn up to Rs 1000 in a single day depending upon performance of the participant. A certificate of participation will also be issued by Manchester University and ISI as recognition of your valued presence. The study will take place during December 2016 to March 2017 for one day in each and every selected GP for 2 hour (approx.) duration. For the convenience of travel, pick- up and drop facilities from convenient locations will be arranged for you. A refreshment packet would be provided to all the participants after the end of each session.

You will not be asked any sensitive question in the sessions. An interesting game will be conducted in each session with you as a participant. The researchers would explain to you clearly the rules of the game. The final result of the game and the data collected in the course of time would be kept confidential and would be used for research purpose only. As an additional precaution, your name and identity will not be disclosed to any one before, during and after any of the sessions. The Professors associated with this work are Sandip Mitra(ISI),Prasenjit Banerjee(MU),Vegard Iversen (MU),Antonio Nicolo(MU) and Kunal Sen(MU).

We firmly believe that you would give your consent to be part of this interesting study and enjoy working with academicians of internationally reputed research organisations. Please do not hesitate to contact me if you have any question.

Regards,

(Sandip Mitra)

Co-PI (MU-ISI Project) ,Contact no : 9830194031

The University of Manchester

Consent form

Respected Sir/Madam,

:

MANCHES

Please fill up the relevant places if you agree to participate in the research study as a subject

- 3. I am convinced that the information provided by me and my identity will be kept completely secret:(yes/no)

I hereby give my consent to be part of the study .

(Name of Participant)

(Name of the Researcher in Charge)

(Signature)

(Signature)

(date)

STATISTICAL

(date)

Incentivised Survey

Design of Incentivised Survey:

(1) Randomly (following the similar process described in the recruitment section above) chosen participants (politicians and non-politicians in a village) were invited to come to a common venue. (2) After the standard protocol (e.g., explaining the survey, written consent etc), the experimenter distributed the incentivised survey-sheet. (3) In the survey, there were two scenarios: (i) Standard dictator game: In an anonymous setting, a dictator was randomly matched with an unknown recipient sitting next room. The dictator was endowed with Rs 1000 (hypothetical) and had to decide how to distribute the endowment with the recipient. (ii) Promise game: Similar to (i) with the following add-on: the dictator had to decide how much to promise and give to the recipient (see the questionnaire in below). (4) Each respondent was asked to rate different combinations of `give' in (i) and `promise and give' in (ii)---ranging from `very socially inappropriate' to `very socially appropriate' on a quantified scale of one to four. (5) Individuals were told that one question would be randomly selected and the person whose answer would match to the modal response of others on that question would be rewarded with Rs. 500 (approx. £5). (6) No communication was allowed. The experimenter made it clear that the survey was anonymous. (7) After completion of the survey, the experimenter collected the survey-sheet from each participant and sent them to an external waiting outside the room. The external then identified who had won Rs 500 reward. (8) Meanwhile, each participant answered a few question related to their demography. (9) The participants left the room one-by-one and the successful participant(s) received the reward, along with the participation fee (Rs. 300), outside the room from the external.

Incentivised Survey Questionnaire

We now ask simply to look at the following scenario.

We will randomly pick answer of one question. If your answer to that question will be the closest match to the overall response, then you will receive Rs 500 as a reward. If more than 10 people give right answers (i.e., will be matched with others' answer), then there will be a lottery so that only 10 lucky people will get the Rs 500 reward.

We ask you now to answer the following questions related to a hypothetical scenario.

Scenario 1

Imagine that two persons A and B participate in a research study. They sit in two separate rooms. They are paired randomly and they didn't know each other before, they don't meet each other in the study, and will never meet each other in the future. In the study, A will make a choice, the experimenter will record this choice, and then both individuals will be informed of the choice and paid money based on the choice made by person A, as well as a small participation fee. Suppose that neither individual will receive any other money for participating in the experiment. The study is anonymous.

Imagine that each pair receives Rs.1000. Person A has then an opportunity to give any amount of his/her Rs. 1000 to person B. A keeps for himself the amount he has not transferred to B. B has no decision to make.

In this scenario,

A's earnings = Rs.1000 - the amount transferred to B.

B's earnings = the amount transferred by A.

This choice will determine how much money each will receive, privately and in cash, at the end of the study.

The table below shows a list the possible choices available to person A. For each of the choices, please indicate whether you believe choosing that option is very socially inappropriate, somewhat socially inappropriate, somewhat socially appropriate, or very socially appropriate. Please tick ' $\sqrt{}$ ' inside one box for each row.

Person A's choice	Very Socially Inappropriate	Somewhat socially inappropriate	Somewhat socially appropriate	Very socially appropriate
Give Rs 0 to Participant B (Person A gets Rs1000, Person B gets Rs 0)				
Give Rs 100 to Participant B (Person A gets Rs 900, Person B gets Rs 100)				
Give Rs 200 to Participant B (Person A gets Rs 800, Person B gets Rs 200)				
Give Rs 300 to Participant B (Person A gets Rs700, Person B gets Rs 300)				
Give Rs 400 to Participant B (Person A gets Rs 600, Person B gets Rs 400)				
Give Rs 500 to Participant B (Person A gets Rs500, Person B gets Rs 500)				
Give Rs 600 to Participant B (Person A gets Rs400, Person B gets Rs 600)				
Give Rs 700 to Participant B (Person A gets Rs 300, Person B gets Rs 700)				
Give Rs 800 to Participant B (Person A gets Rs 200, Person B gets Rs 800)				
Give Rs 900 to Participant B (Person A gets Rs 100, Person B gets Rs 900)				
Give Rs 1000 to Participant B (Person A gets Rs 0, Person B gets Rs 1000)				

Alternatively, you can indicate any other option you believe choosing that option by person A is very socially appropriate: Give Rs._____

Scenario 2

Imagine that two persons A and B participate in a research study. They sit in two separate rooms. They are paired randomly and they didn't know each other before, they don't meet each other in the study, and will never meet each other in the future. In the study, A will make a choice, the experimenter will record this choice, and then both individuals will be informed of the choice and paid money based on the choice made by person A, as well as a small participation fee. Suppose that neither individual will receive any other money for participating in the experiment. The study is anonymous.

Imagine that each pair receives Rs.1000. Person A has then an opportunity to give any amount of his/her Rs. 1000 to person B. A keeps for himself the amount he has not transferred to B. B has no decision to make. In this scenario,

A's earnings = Rs.1000 – the amount transferred to B.

B's earnings = the amount transferred by A.

Before making the actual decision, the person A is asked to write the amount that he intends to give to B in a sealed envelope. The researcher then gives the envelope to person B in the next room and B sees A's promised amount.

Then A actually decides how much to give. His actual decision may or may not match with the promised amount. No one else can see A's actual decision. Person A leaves first and then B leaves the study centre. This choice will determine how much money each will receive, privately and in cash, at the end of the study.

The table below shows a list the possible choices available to person A. For each of the choices, please indicate whether you believe choosing that option is very socially inappropriate, somewhat socially inappropriate, somewhat socially appropriate, or very socially appropriate. Please tick ' $\sqrt{}$ ' inside one box for each row.

Person A's choice	Very Socially	Somewhat socially	Somewhat socially	Very socially
	Inappropriate	inappropriate	appropriate	appropriate
Promise Rs. 0 and give Rs 0 to Person B				
Promise Rs. 0 and give more than Rs 0 to Person B				
Promise Rs 500 and give Rs 500 to Person B				
Promise Rs 500 and give Rs 0 to Person B				
Promise Rs 500 and give more than Rs 500 to Person B				
Promise Rs 500 and less than Rs 500 to Person B				
Promise Rs. 1000 and give Rs 1000 to Person B				
Promise Rs. 1000 and give Rs 0 to Person B				
Promise Rs 1000 and give Rs 500 to Person B				
Promise Rs 1000 and give in between Rs 500 and Rs 900 to Person B				
Promise more than Rs 500 and give the promised amount to Person B				
Promise any amount in between Rs 200 and Rs 400 and give the promised amount to Person B				
Promise any amount in between Rs 200 and Rs 400 and give less than the promised amount to Person B				

Alternatively, you can indicate any other option you believe choosing by person A that option is very socially appropriate: Promise Rs: ______ and give Rs._____
Additional Information

1. Male () Female ()
2. Age:
3. Household main occupation (coded):
4. What is your main occupation, if any (coded):
5. What jati do you belong to?
6. Religion: Hindu () Muslim () Other ()
7. How many years of education have you completed? years
If you have an educational degree, which is the highest degree?
 How much did your household earn in the last year?
9. Are you presently an elected representative in your Gram Panchayat? YES NO
a. In what position?
 b. Is this a reserved seat? YES NO If, YES, which reservation category
c. Is this your first period as an elected representative? YES NO
If YES, how many periods have you served?
10. Has any other member of your household or near family been a GP representative? YES NO

11. If a current representative, are you planning to stand for the next election? YES NO